SECURITIES AND EXCHANGE COMMISSION

FORM SC 13D/A

Schedule filed to report acquisition of beneficial ownership of 5% or more of a class of equity securities [amend]

Filing Date: **2002-07-02 SEC Accession No.** 0000065195-02-000013

(HTML Version on secdatabase.com)

FILED BY

MESTEK INC

CIK:65195| IRS No.: 250661650 | State of Incorp.:PA | Fiscal Year End: 1231

Type: SC 13D/A

SIC: 3585 Air-cond & warm air heatg equip & comm & indl refrig equip

Business Address 260 N ELM ST WESTFIELD MA 01085 4135689571

SUBJECT COMPANY

CARECENTRIC INC

CIK:896157| IRS No.: 223209241 | State of Incorp.:DE | Fiscal Year End: 1231

Type: SC 13D/A | Act: 34 | File No.: 005-44440 | Film No.: 02695049

SIC: 7372 Prepackaged software

Mailing Address 2625 CUMBERLAND PARK SUITE 310 ATLANTA GA 30339

Business Address 2625 CUMBERLAND PARK SUITE 310 ATLANTA GA 30339 678-264-4400

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 13D/A

Under the Securities Exchange Act of 1934 (Amendment No. 4)

CareCentric, Inc.

(Name of Issuer)

Common Stock, \$0.001 Par Value

(Title of Class of Securities)

828654301 (CUSIP Number)

J. Nicholas Filler, Esq., Mestek, Inc., 260 North Elm Street, Westfield, MA 01085 (413) 564-5514

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

July 1, 2002

(Date of Event which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition which is the subject of this Schedule 13D, and is filing this schedule because of Rule 13d-1(b)(3) or (4), check the following box .

Note: Six copies of this statement, including all exhibits, should be filed with the Commission. See Rule 13d-1(a) for other parties to whom copies are to be sent.

*The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the

CUSIP No. 828654301								
1	Name of Reporting Person: Mestek, Inc. 25-0661650							
2	Check the Appropriate Box if a Member of a Group	(a) (b)	[X]					
3	SEC Use Only			_				
4	Source of Funds							
5	Check Box if Disclosure of Legal Proceedings is Required Pursuant to Items 2(d) or 2(e)		[]					
6	Citizenship or Place of Organization Pennsylvania							
7	Sole Voting Power 10,890,396 (1)(2)							
8	Shared Voting Power			-				
9	Sole Dispositive Power 10.490.396 (1)(2)			-				

Notes).

10	Shared Dispositive Power	0
	Aggregate Amount Beneficially Owned by Each Report 10,890,396 (1)(2	2)
12	Check Box if the Aggregate Amount in Row (11) Exc	ludes Certain Shares
	Percent of Class Represented by Amount in Row (11)	3)
	Type of Reporting Person	CO

SEE INSTRUCTIONS BEFORE FILLING OUT

- (1)Includes (a) warrant to purchase 400,000 shares of the Company's Common Stock at a per share price of \$1.00 received on July 1, 2002 in exchange of a prior warrant, pursuant to the closing of a Secured Convertible Credit Facility (the "Mestek Credit Facility") between Mestek, Inc. ("Mestek") and CareCentric, Inc. ("CareCentric") dated as of July 1, 2002, which warrant expires on June 15, 2004, (b) a warrant to purchase 490,396 shares of the Company's Common Stock at a per share price of \$1.00 on July 1, 2002 received in exchange of a prior warrant, pursuant to the closing of the Mestek Credit Facility dated as of July 1, 2002, which warrant expires on June 15, 2004, (c) 4,000,000 shares of the Company's Common Stock issuable upon conversion of outstanding indebtedness under the Mestek Credit Facility, and (d) 6,000,000 shares of the Company's Common Stock issuable upon conversion of 5,600,000 shares of the Company's Series B Preferred Stock. As of the date of this Schedule, Mestek, Inc. does not have any issued shares of the Company's Common Stock.
- (2) Includes 5,600,000 shares of Series B Preferred Stock, for which the right to vote on matters as to which all shareholders may vote have been transferred to John E. Reed pursuant to a voting agreement dated March 29, 2002. If the Series B Preferred Stock is converted, the Voting Agreement does not provide for transfer of the voting rights of the Common Stock, received in conversion, to Mr. Reed. Certain voting rights issuable upon exercise of certain options, a warrant to purchase 104,712 shares of the Company's Common Stock at a per share price of \$2.51 received on July 12, 2000, and Series C Preferred Stock have all

been cancelled.

(3) Calculated as percentage of currently issued Common Stock plus only Mestek, Inc. aggregate amount beneficially owned and voting power if all Mestek derivative securities are exercised or converted.

Mestek hereby amends and supplements its Schedule 13D as filed on March 17, 2000, and as amended by Amendment No. 1 filed on June 14, 2000, Amendment No. 2 filed on July 18, 2000 and Amendment No. 3 filed on December 31, 2001 (as amended, the ASchedule 13D@), with respect to the Common Stock, par value \$.001 per share (the ACommon Stock@), of CareCentric, Inc., a Delaware corporation (the ACompany@), formerly known as Simione Central Holdings, Inc.

Item 3. Source and Amount of Funds or Other Consideration

Item 3 of the Schedule 13D is hereby amended and supplemented as follows:

Pursuant to a Secured Convertible Credit Facility and Security Agreement, dated as of July 1, 2002 (the AMestek Credit Facility@), Mestek has extended loans to the Company in the principal amount of \$4,000,000 (the AMestek Indebtedness@). The Mestek Indebtedness is currently convertible into Common Stock at a conversion price of \$1.00 per share. As a result, 4,000,000 shares of Common Stock are currently issuable, upon conversion, to Mestek under the Mestek Credit Facility. In addition to refinancing existing debt, additional advances under the Mestek Credit Facility are from working capital of Mestek.

The Mestek Credit Facility contemplates the capitalization of interest over the first 24 months of the Mestek Credit Facility. Therefore, as of June 30, 2004, it is contemplated that the Mestek Credit Facility will have a stated principal amount outstanding of \$4,500,000, and as a result, 4,500,000 shares of Common Stock would be then issuable, upon conversion, to Mestek under the Mestek Credit Facility, if no principal repayments, debt payments or conversions have taken place prior thereto.

The Mestek Credit Facility refinances indebtedness due to Mestek under that certain Participation Agreement dated December 31, 2001 between Mestek, the Company and John E. Reed ("Mr. Reed"). In that Participation Agreement Mestek had extended loans to the Company in the principal amount of \$1,000,000 in participation with Mr. Reed who had entered into a Secured Convertible Credit Facility and Security Agreement dated June 12, 2000, which has been amended and restated as of July 1, 2002 (as amended and restated, the "Reed Credit Facility"). The Participation Agreement has no further effect.

The Mestek Credit Facility was issued in consideration of the refinancing of various other indebtedness of the Company to Mestek as well as the cancellation by Mestek of (a) Mestek's right subject to certain conditions

to purchase up to approximately 378,295 shares (the "Option Shares") of Company Common Stock issuable pursuant to an Option Agreement dated as of March 7, 2000 between Mestek and the Company (the "Option Agreement"), (b) 850,000 shares of Series C Preferred Stock by the Company to Mestek and (c) a warrant to purchase 104,712 shares of the Company's Common Stock at a per share price of \$2.51 received on July 12, 2000 in connection with Mestek's guaranty of the Company's obligations under a line of credit from Wainwright Bank & Trust Company and which would have expired on July 12, 2003, all of the foregoing having been cancelled by Mestek pursuant to the Mestek Credit Facility.

In addition, as partial consideration for the Mestek Credit Facility, Mestek has exchanged with the Company (a) a warrant to purchase 400,000 of the Company's Common Stock at a per share purchase price of \$10.875 received on March 7, 2000 and which would have expired on March 7, 2003 for a warrant to purchase 400,000 shares of the Company's Common Stock at a per share price of \$1.00 exchanged on July 1, 2002 and with an expiration date of June 15, 2004, and (d) a warrant to purchase 490,396 shares of the Company's Common Stock at a per share price of \$3.21 received on June 12, 2000 and which would have expired on June 12, 2003, for a warrant to purchase 490,396 shares of the Company's Common Stock at a per share price of \$1.00 exchanged on July 1, 2002, and with an expiration date of June 15, 2004. The prior warrants have no further effect.

Item 4. Purpose of Transaction

Item 4 of the Schedule 13D is hereby amended and supplemented as follows:

Pursuant to the Mestek Credit Facility, Mestek has extended loans to the Company in the principal amount of \$4,000,000. The Mestek Indebtedness is currently convertible into Common Stock at a conversion price of \$1.00 per share. As a result, 4,000,000 shares of Common Stock are currently issuable, upon conversion, to Mestek under the Mestek Credit Facility.

The Mestek Credit Facility contemplates the capitalization of interest over the first 24 months of the Mestek Credit Facility and as a result, as of June 30, 2004, it is contemplated that the Mestek Credit Facility will have a stated principal amount outstanding of \$4,500,000, and as a result, 4,500,000 shares of Common Stock would be then issuable, upon conversion, to Mestek under the Mestek Credit Facility, if no principal repayments, debt payments or conversions have taken place prior thereto.

The Mestek Credit Facility refinances indebtedness due to Mestek under that certain Participation Agreement dated December 31, 2001 between Mestek, the Company and John E. Reed ("Mr. Reed"). In that Participation Agreement Mestek had extended loans to the Company in the principal amount of \$1,000,000 in participation with Mr. Reed who had entered into a Secured Convertible Credit Facility and Security Agreement dated June 12, 2000, which has been amended and

restated as of July 1, 2002 (as amended and restated, the "Reed Credit Facility") and who is Chairman and Chief Executive Officer of Mestek, Inc.

The Mestek Credit Facility was issued in consideration of the refinancing of various other indebtedness of the Company to Mestek as well as the cancellation by Mestek of (a) Mestek's right subject to certain conditions to purchase up to approximately 378,295 shares (the "Option Shares") of Company Common Stock issuable pursuant to an Option Agreement dated as of March 7, 2000 between Mestek and the Company (the "Option Agreement"), (b) 850,000 shares of Series C Preferred Stock by the Company to Mestek and (c) a warrant to purchase 104,712 shares of the Company's Common Stock at a per share price of \$2.51 received on July 12, 2000 in connection with Mestek's guaranty of the Company's obligations under a line of credit from Wainwright Bank & Trust Company and which would have expired on July 12, 2003, all of the foregoing having been cancelled by Mestek pursuant to the Mestek Credit Facility.

In addition, as partial consideration for the Mestek Credit Facility, Mestek has exchanged with the Company (a) a warrant to purchase 400,000 of the Company's Common Stock at a per share purchase price of \$10.875 received on March 7, 2000 and which would have expired on March 7, 2003 for a warrant to purchase 400,000 shares of the Company's Common Stock at a per share price of \$1.00 exchanged on July 1, 2002 and with an expiration date of June 15, 2004, and (d) a warrant to purchase 490,396 shares of the Company's Common Stock at a per share price of \$3.21 received on June 12, 2000 and which would have expired on June 12, 2003, for a warrant to purchase 490,396 shares of the Company's Common Stock at a per share price of \$1.00 exchanged on July 1, 2002, and with an expiration date of June 15, 2004.

The Mestek Credit Facility provides Mestek with a right to convert outstanding indebtedness, if any, into shares of Common Stock of the Company. Any outstanding indebtedness would be convertible into a whole number of shares of Common Stock determined by dividing the dollar amount of indebtedness to be converted by the conversion price. The conversion price is currently \$1.00 per share and is subject to certain adjustment pursuant to an anti-dilution mechanism. This mechanism would, with certain exceptions, lower the conversion price in the event of any subsequent issuance by the Company of Common Stock or options to purchase Common Stock at a per share price of less than \$1.00. The mechanism also provides for adjustments for stock splits, stock dividends, or share reclassifications. \$4,000,000 of indebtedness has been issued and therefor 4,000,000 shares of Common Stock are immediately issuable, upon conversion, to Mestek under the Credit Facility should Mestek elect to convert.

The Mestek Credit Facility provides Mestek with registration rights for any shares to be issued to Mestek upon conversion of Mestek indebtedness whereby the Company will upon written request of Mestek and no earlier than six months following execution of the Mestek Credit Facility, effect up to three registrations of Common Stock received by Mestek upon any conversion of Mestek indebtedness pursuant to the Mestek Credit Facility. Pursuant to the Mestek

Credit Facility, the Company agrees to use its best efforts to utilize shelf registration under Rule 415 of the Securities Act. Similarly, if the Company effects registration on its own accord for any of its equity securities, Mestek will have the right to request registration of additional shares, subject to certain senior registration rights, referenced in the Mestek Credit Facility. The Company's obligation to register the Common Stock ceases if Mestek is entitled to sell the shares pursuant to Rule 144(k) of the Securities Act.

Other than as described in this Item 4, Mestek does not have any present plans or proposals which relate to or would result in (1) the acquisition by any person of additional securities of the Company or the disposition of securities of the Company (2) an extraordinary corporate transaction, such as a merger, reorganization or liquidation, involving the Company or any of its subsidiaries, (3) a sale or transfer of a material amount of the assets of the Company or of any of its subsidiaries, (4) any material change in the capitalization or dividend policy of the Company, (5) any other material change in the Company's business or corporate structure, (6) changes to the Company's charter, by-laws, or instruments corresponding thereto or other actions which may impede the acquisition of control of the Company by any person, (7) causing a class of securities of the Company to be delisted from a national securities exchange or cease to be authorized to be quoted in any inter-dealer quotation system of a registered national securities association, (8) a class of equities securities of the Company becoming eliqible for termination of registration pursuant to Section 12(g)(4) of the Act, or (9) any action similar to any of those enumerated above.

As additional consideration for the issuance of the Mestek Credit Facility, the Company submitted to its shareholders and the shareholders voted to amend the designations of Series B Preferred Stock (of which Mestek is the only holder of all currently issued and outstanding shares), to provide that Series B Stock, which was previously not convertible, may be converted into shares of Common Stock of the Company. Upon filing of a statement of designations, rights and preferences with the Delaware Secretary of State each share of Series B Preferred Stock is convertible into a whole number of shares of Common Stock determined by dividing the original purchase price of \$1.07 by the conversion price. The conversion price is currently \$1.00 per share and is subject to adjustment pursuant to an anti-dilution mechanism. This mechanism would, with certain exceptions, lower the conversion price in the event of any subsequent issuance by the Company of Common Stock at a per share purchase price of less than \$1.00. The mechanism also provides for adjustments for stock splits, stock dividends or share reclassifications. The 5,600,000 shares of Series B Preferred Stock are currently convertible into 6,000,000 shares of Common Stock. In all other respects, the designations conditions, limitations and terms of the Series B Preferred Stock remain the same as prior to the Mestek Credit Facility transaction. Pursuant to a Voting Agreement between Mestek, Inc. and John E. Reed, dated March 29, 2002, Mestek has transferred to Mr. Reed the right to vote on matters upon which all shareholders are entitled to vote with respect to its 5,600,000 shares of Series B Preferred Stock only. If Series B

Preferred Stock is converted, the voting agreement does not provide for transfer of the voting rights of the Common Stock, received in conversion, to Mr. Reed.

Notwithstanding the foregoing, Mestek reserves the right to purchase additional securities of the Company, and dispose of all or a portion of its holdings of securities of the Company, or change its intentions with respect to any of the matters referred to in this Item 4.

Item 5. Interest in Securities of the Issuer

Item 5(c) is amended to read as follows:

- (c) See Items 3 and 4. No other transactions in the Company's Common Stock have been effected by the person named in Item 2 within the last sixty days
- Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer

Item 6 is supplemented as follows:

All contracts, arrangements, understandings or relationships with respect to the securities of the Company between Mestek and any other person are described in Item 4 above, which is incorporated herein by reference, and the Secured Convertible Credit Facility and Security Agreement, dated July 1, 2002, between Mestek, Inc. and CareCentric, Inc. which is attached as Exhibit 99.3 and which is incorporated herein by reference, and the Voting Agreement between John E. Reed and Mestek, Inc. dated March 29, 2002, which is attached hereto as Exhibit 99.4 and which is incorporated herein by reference, and the Warrant for 400,000 shares which is attached hereto as Exhibit 99.5 and which is incorporated herein by reference and the Warrant for 490,396 shares which is attached hereto as Exhibit 99.6 and which is incorporated herein by reference.

Item 7. Material to be Filed as Exhibits

In addition to Exhibits previously filed, the following Exhibits to this Amendment No. 4 of Schedule 13D are:

- EX-99.3 Secured Convertible Credit Facility and Security Agreement, dated as of July 1, 2002, between Mestek, Inc. and CareCentric, Inc. is attached hereto as Exhibit 99.3.
- EX-99.4 Voting Agreement between John E. Reed and Mestek, Inc. dated as March 29, 2002 as attached hereto as Exhibit 99.4.

Exhibit 99.5.

EX-99.6 Warrant for 490,396 shares which is attached hereto as Exhibit 99.6.

Warrant for 400,000 shares which is attached hereto as

Signatures.

EX-99.5

After reasonable inquiry each of the undersigned certifies that to the best of his knowledge and belief the information set forth in this statement is true, complete and correct.

MESTEK, INC.

By: /S/ STEPHEN M. SHEA
----Stephen M. Shea

June 26, 2002

Date

WARRANT

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR ANY STATE SECURITIES LAWS AND, UNLESS SO REGISTERED, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND APPLICABLE STATE SECURITIES LAWS.

Warrant to purchase 490,396 shares of the \$0.001 par value common stock of CareCentric, Inc. (subject to adjustment)

WARRANT TO PURCHASE COMMON STOCK OF CARECENTRIC, INC.

This certifies that, for value received, Mestek, Inc., or its successors or assigns (the "Holder"), is entitled, subject to the terms set forth below, to purchase CareCentric, Inc. (the "Company") up to 490,396 shares of the \$0.001 par value common stock of the Company ("Common Stock"), as the Company is constituted on the 1st day of July, 2002 (the "Warrant Issue Date"), upon surrender of this certificate at 2625 Cumberland Parkway, Suite 310, Atlanta Georgia, or such other place as the Company may designate in writing to the Holder, and the simultaneous payment therefor in lawful money of the United States of America of the Exercise Price (as hereinafter defined). The number, character and Exercise Price of such shares are subject to adjustment as provided herein. The term "Warrant" as used herein shall include this certificate, the securities represented by this certificate and any warrants delivered in substitution or exchange for this certificate as provided herein.

1. Term of Warrant. Subject to the terms and conditions set forth herein, this Warrant shall be exercisable, in whole or in part, during the period of time (the "Exercise Period") commencing on the Warrant Issue Date and ending at 5:00 p.m. on June 15, 2004, and shall be void thereafter.

Exercise Price. The price at which the Holder may exercise this Warrant (the "Exercise Price") shall be \$1.00 per share, subject, however, to adjustments as provided in Section 9 hereof.

- 3. Vesting of Warrant. Effective as of the Warrant Issue Date, the Warrant shall be fully vested and exercisable, and the Holder shall have the fully vested right to purchase 400,000 shares of Common Stock pursuant to the terms and conditions of this Warrant.
- 4. Exercise of the Warrant. The purchase rights represented by this

Warrant are exercisable by the Holder, in whole or in part, at any time, and from time to time during the Exercise Period, by the Holder's surrender of this Warrant at 2625 Cumberland Parkway, Suite 310, Atlanta Georgia, or such other place as the Company may designate in writing to the Holder, and the simultaneous payment therefor in lawful money of the United States of America of the Exercise Price in immediately available funds. This Warrant shall be deemed exercised on the date immediately prior thereto, and the Holder shall be entitled to receive the shares of Common Stock and be treated for all purposes as the holder of record of such shares as of the close of business on such date. As promptly as practicable, but in no event later than ten (10) business days thereafter, the Company shall issue and deliver, at its sole cost and expense, to the person or persons entitled to receive the same a certificate or certificates for the number of shares issuable upon such exercise. In the event that this Warrant is exercised in part, the Company, at its sole cost and expense, shall execute and deliver a new warrant of like tenor as this Warrant, exercisable for the remaining number of shares for which this Warrant may then be exercised, and shall cancel this Warrant only upon issuance of such new warrant. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant, and in lieu thereof, the Company shall make a cash payment to the Holder equal to the Exercise Price multiplied by such fraction.

5. Rights as a Stockholder. The Holder shall not be entitled to vote, receive dividends or be deemed to be the owner of record of the shares of Common Stock to which this Warrant relates unless and until the Holder exercises this Warrant, and then the Holder shall enjoy such rights only to the extent of such exercise.

6. Transfer of Warrant.

- (a) Warrant Register. The Company will maintain a register (the "Warrant Register") maintaining the names and addresses of the Holder or Holders. Any Holder of this Warrant or any portion thereof may change its address as shown on the Warrant Register by written notice to the Company requesting such change. Any notice or written communication required or permitted to be given to the Holder may be delivered or given by mail to such Holder as shown on the Warrant Register and at the address shown on the Warrant Register. Until this Warrant is transferred on the Warrant Register of the Company, the Company may treat the Holder as shown on the Warrant Register as the absolute owner of this Warrant for all purposes, notwithstanding any notice to the contrary.
- (b) Warrant Agent. The Company may, by written notice to the Holder, appoint an agent for the purpose of maintaining the Warrant Register referred to in Section 6(a) above, issuing the Common Stock or other securities then issuable upon the exercise of this Warrant, exchanging this Warrant, replacing

this Warrant, or any or all of the foregoing. Thereafter, any such registration, issuance, exchange, or replacement, as the case may be, shall be made at the office of such agent.

- (c) Transferability of Warrant. This Warrant may not be transferred or assigned (i) except in its entirety (other than transfers to subsidiaries or affiliates of Mestek, Inc.) and (ii) without compliance with all applicable federal and state securities laws by the transferor and the transferee (including delivery of investment representation letters reasonably satisfactory to the Company, if such are requested by the Company), and then only against receipt of an agreement of the transferee to comply with the provisions of this Section 6(c) with respect to any resale or other disposition of this Warrant.
- (d) Exchange of Warrant upon a Transfer. On surrender of this Warrant for exchange, properly endorsed and subject to the provisions of this Warrant with respect to compliance with the Act and with the limitations on assignments and transfers contained in this Section 6, the Company at its expense shall issue to or on the order of the Holder a new Warrant or Warrants of like tenor, in the name of the Holder or as the Holder (on payment by the Holder of any applicable transfer taxes) may direct, for the number of shares issuable upon exercise hereof.
- (e) Compliance with Securities Laws.
- (i) The Holder of this Warrant, by acceptance hereof, acknowledges that this Warrant and the shares of Common Stock to be issued upon exercise hereof are being acquired solely for the Holder's own account and not as a nominee for any other party, and for investment, and that the Holder will not offer, sell or otherwise dispose of this Warrant or any shares of the Common Stock to be issued upon exercise hereof except under circumstances that will not result in a violation of the Securities Act of 1933 or any state securities laws. Upon exercise of this Warrant, the Holder shall, if requested by the Company, confirm in writing, in a form satisfactory to the Company, that the shares of Common Stock so purchased are being acquired solely for the Holder's own account and not as a nominee for any other party, for investment, and not with a view toward distribution or resale.
- (ii) This Warrant and all shares of Common Stock issued upon exercise hereof or conversion thereof shall be stamped or imprinted with a legend in substantially the following form (in addition to any legend required by state securities laws): THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR ANY STATE SECURITIES LAWS AND, UNLESS SO REGISTERED, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND APPLICABLE STATE SECURITIES LAWS.

- 7. Reservation of Stock. The Company covenants that during the Exercise Period, the Company will reserve from its authorized and unissued shares of treasury Common Stock a sufficient number of shares to provide for the issuance of Common Stock upon the exercise of the Warrant and, from time to time, will take all steps necessary to amend its certificate of incorporation (the "Certificate") to provide sufficient authorized reserved shares of Common Stock issuable upon exercise of the Warrant. The Company further covenants that all shares that may be issued upon exercise of the rights represented by this Warrant and payment of the Exercise Price, all as set forth herein, will be free from all taxes, liens and charges in respect of the issue hereof (other than taxes in respect of any transfer occurring contemporaneously or otherwise specified herein). The Company agrees that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of executing stock certificates to execute and issue the necessary certificates for shares of Common Stock upon the exercise of this Warrant.
- 8. Merger; Sale of Assets and other Fundamental Corporate Changes. If at any time during the Exercise Period there shall be a sale of all or substantially all of the Company assets, or a merger, consolidation or reorganization of the Company in which the Company is not the surviving entity, or other transaction in which the shares of the Company are converted into shares of another entity, the Company shall provide the Holder with written notice thereof not less than 30 calendar days prior to the consummation of such event and an opportunity to exercise this Warrant prior to the consummation of such event.
- 9. Adjustments to Exercise Price for Certain Diluting Issuances. Splits and Combinations. The Exercise Price shall be subject to adjustment from time to time as follows:
- (a) Special Definitions. For purposes of this Section 9, the following definitions apply:
- (i) "Options" shall mean rights, options, or warrants to subscribe for, purchase or otherwise acquire either Common Stock or Convertible Securities (defined below).

"Convertible Securities" shall mean any evidences of indebtedness, shares (other than Common Stock) or other securities convertible into or exchangeable for Common Stock.

"Additional Stock" shall mean all shares of Common Stock issued by the Company after the Warrant Issue Date, and all shares of Common Stock issuable pursuant to Options and Convertible Securities issued by the Company after the Warrant Issue Date, other than shares of Common Stock for which adjustment of the Exercise Price is made pursuant to Section 9(d) or 9(e) below.

- (b) Adjustments. If the Company shall issue, after the Warrant Issue Date, any Additional Stock without consideration or for a consideration per share less than the Exercise Price in effect immediately prior to the issuance of such Additional Stock, the Exercise Price in effect immediately prior to each such issuance shall forthwith be adjusted downward to a price equal to the price paid per share for such Additional Stock.
- (c) Determination of Consideration. For purposes of this Section 9, the consideration received by the Company for the issuance of any Additional Stock shall be computed as follows:
 - (i) Cash and Property. Such consideration shall:
- (A) insofar as it consists of cash, be computed at the aggregate amount of cash received by the Company excluding amounts paid or payable for accrued interest or accrued dividends;
- (B) insofar as it consists of property other than cash, be computed at the fair value thereof at the time of such issue, as determined in good faith by the Company's Board of Directors; and
- (C) in the event Additional Stock is issued together with other shares or securities or other assets of the Company for consideration which covers both, be the proportion of such consideration so received, computed as provided above, as determined in good faith by the Company's Board of Directors.
- (ii) Options and Convertible Securities. The consideration per share received by the Company for Additional Stock deemed to have been issued pursuant to this Section 9 relating to Options and Convertible Securities shall be determined by dividing:
- (A) the total amount, if any, received or receivable by the Company as consideration for the issuance of such Options or Convertible Securities, plus the minimum aggregate amount of additional consideration (as set forth in the instruments relating thereto, without regard to any provision contained therein designed to protect against dilution) payable to the Company upon the exercise of such Options or the conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities, by
- (B) the maximum number of shares of Common Stock (as set forth in the instruments relating thereto, without regard to any provision contained therein designed to protect against dilution) issuable upon the exercise of such Options or conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible

Securities.

- (d) Adjustments to Exercise Price for Stock Dividends and for Combinations Subdivisions of Common Stock. In the event that the Company at any time or from time to time after the Warrant Issue Date shall declare or pay any dividend on the Common Stock payable in Common Stock or in any right to acquire Common Stock for no consideration, or shall effect a subdivision of the outstanding shares of Common Stock into a greater number of shares of Common Stock (by stock split, reclassification or otherwise), or in the event the outstanding shares of by reclassification or Common Stock shall be combined or consolidated, otherwise, into a lesser number of shares of Common Stock, then the Exercise Price in effect immediately prior to such event shall, concurrently with the effectiveness of such event, be proportionately decreased or increased, appropriate. In the event that the Company shall declare or pay any dividend on the Common Stock payable in any right to acquire Common Stock for consideration, then the Company shall be deemed to have made a dividend payable in Common Stock in an amount of shares equal to the maximum number of shares issuable upon exercise of such rights to acquire Common Stock.
- Adjustments for Reclassification and Reorganization. If the Common shall be changed into the same or a different number of shares of any classes of stock, whether by capital class or reorganization, reclassification or otherwise (other than a subdivision or combination of shares provided for in Section 9(d) above) the applicable Exercise Price then in effect concurrently with the effectiveness of such reorganization or reclassification, be proportionately adjusted so that the shares of Common Stock issued upon exercise hereof shall be convertible into, in lieu of the number of shares of Common Stock which the Holder would otherwise have been entitled to a number of shares of such other class or classes of stock equivalent to the number of shares of Common Stock that would have been subject to receipt by the Holder upon conversion of shares of the Common Stock immediately before that change.
- 10. Registration Rights. The Company covenants and agrees that the Holder shall have registration rights with respect to this Warrant as follows:
 - 10.1 Definitions. For purposes of this Section 10:
- (a) The term "Form S-3" means such form under the Securities Act of 1933, as amended ("Securities Act"), as in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.

- (b) The term "Form S-4" means such form under the Securities Act as in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC for corporate combinations and exchange offers which peanuts inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
- (c) The term "Holder" means the Holder or any permitted transferee or assignee thereof.
- (d) The term "Person" means an individual, a corporation, a limited liability company, a partnership, an association, a trust or any other entity organization, including a governmental entity.
- (e) This terms "register," "registered." and "registration" refer to a registration effected by preparing and filing a registration statement or similar document in compliance with the Securities Act, and the declaration or ordering of effectiveness of such registration statement or document.

The term "Registrable Shares" means (i) the Company's Common Stock issuable or issued upon exercise of this Warrant, and (ii) any Common Stock or other securities issued or issuable in respect of shares referenced in (i) above, upon any stock split, stock dividend, recapitalization, or similar event; excluding in all cases, however, any Registrable Shares sold by a Person in a transaction in which such Person's rights under this Section 10 are not assigned.

- (g) The term "SEC" means the Securities and Exchange Commission.
- (h) The term "Subsidiary" means, with respect to any Person, any corporation, limited liability company, or partnership of which such Person owns, either directly or through its subsidiaries or affiliates, more than fifty percent (50%) of (i) the total combined voting power of all classes of voting securities in the case of a corporation or (ii) the capital or profit interests therein in the case of a partnership.
- 10.2 Request for Registration. Upon request of the Holder, the Company will use its best efforts to file within 45 days of a request from the Holder a registration statement with the SEC (utilizing Form S-3 or a successor form thereto and Rule 415 to the extent available) to register Registrable Shares as requested by the Holder. The Company shall not be required to file more than three such registration statements (excluding any registration statement which is delayed pursuant to Section 10.4(e) below and through which the Holder is unable to register eighty percent (80%) or more of the amount of Registrable Shares that the Holder originally requested to register in such registration statement), and no such filing shall be made prior to the date which is six (6) months after the Warrant Issue Date.
 - 10.3 Company Registration. If the Company at any time proposes to register

an offering of its securities under the Securities Act, either for its own account or for the account of or at the request of one or more Persons holding securities of the Company, the Company will:

- (a) give written notice thereof to the Holder (which shall include a list of the jurisdictions in which the Company intends to attempt to qualify such securities under the applicable blue sky or other state securities laws) within 10 days of its receipt of a request from one or more Persons holding securities of the Company to register securities, or from its decision to effect a registration of securities for its own account, whichever first occurs; and
- (b) use its best efforts to include in such registration and in any underwriting involved therein, all the Registrable Shares specified in a written request by the Holder made within 30 days after receipt of such written notice from the Company, except as set forth in Section 10.4(e) below and subject to the currently existing piggyback rights referenced in Section 10.10.
- 10.4 Obligations of the Company. If and whenever pursuant to the provisions of this Section 10 the Company effects registration of Registrable Shares under the Securities Act and state securities laws, the Company shall:
- (a) Prepare and file with the SEC a registration statement with respect to such securities and use its best efforts to cause such registration statement to become and remain effective for a period not to exceed two years after the filing (but which period shall be extended by the duration of any delay periods under clause (e) below);
- (b) Use its best efforts to register or qualify the securities covered by such registration statement under the securities or blue sky laws of such jurisdictions as the Holder shall reasonably request, and do any and all other acts and things which may be necessary or advisable (in the reasonable opinion of the Holder) to enable the Holder to consummate the disposition thereof; provided, however, that in no event shall the Company be obligated to qualify to do business in any jurisdiction where it is not now so qualified or to take any action which would subject it to the service of process in suits other than those arising out of the offer or sale of the securities covered by such registration statement in any jurisdictions where it is not now so subject;
- (c) As promptly as practicable prepare and file with the SEC such amendments and supplements to any registration statement and prospectus used pursuant to or in connection with this Agreement as may be necessary to keep such registration statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement or for such shorter period as may be required herein; and

- (d) Furnish to the Holder such number of conformed copies of its registration statement and of each such amendment and supplement thereto (in each case including all exhibits, such number of copies of the prospectus comprised in such registration statement (including each preliminary prospectus and any summary prospectus), in conformity with the requirements of the Securities Act), and such other related documents as the Holder may reasonably request in order to facilitate the disposition of the Registrable Shares to be registered.
- (e) Anything in this Agreement to the contrary notwithstanding:
 - (i) The Company may defer the filing ("Filing") of any registration statement or suspend the use of a prospectus under a currently effective registration statement under this Agreement at its discretion for "Good Cause." "Good Cause" means either if (1) the Company is engaged in active negotiations with respect to the acquisition of a "significant subsidiary" as defined in Regulation S-X promulgated by the SEC under the Securities Exchange Act of 1934, as amended ("Exchange Act") and the Securities Act which would in the opinion of counsel for the Company be required to be disclosed in the Filing; or (2) in the opinion of counsel for the Company, the Filing would require the inclusion therein of certified financial statements other than those in respect of the Company's most recently ended full fiscal year and any preceding full fiscal year, and the Company may then, at its option, delay the imposition of its registration obligations hereof until the earlier of (A) the conclusion or termination of such negotiations, or the date of availability of such certified financial statements, whichever is applicable, or (B) 60 days from the date of the registration request.
- (ii) In the event the Company has deferred a requested Filing, pursuant to the preceding paragraph, such deferral period shall end if the Company registers shares for resale by another stockholder of the Company. In the event the Company undertakes an underwritten public offering to issue the Company securities for cash during any period in which a requested Filing has been deferred or if the registration of which the Company gives notice under Section 10.3(a) is for an underwritten public offering to issue the Company securities for cash, the Company shall include the Registrable Shares in such underwritten offering subject to (A) the right of the managing underwriters to object to including such shares, (B) Section 10.10, and (C) the condition that the Holder selling Registrable Shares in such underwritten offering shall cooperate in the registration process in all material respects, including execution by the Holder of the underwriting agreement agreed to by the Company and the underwriters.

- (iii) If the managing underwriter elects to limit the number or amount of securities to be included in any registration referenced in the preceding paragraph or in Section 10.3(a), all Persons holding securities of the Company (including the Holder) who hold registration rights and who have requested registration (collectively, the "Security Holders") shall, subject to Section 10.10 hereof, participate in the underwritten public offering pro rata based upon the ratio of the total number or amount of securities to be offered in the offering to the total number or amount of securities held by each Security Holder (including the number or amount of securities which each such Security Holder may then be entitled to receive upon the exercise of any option or warrant, or the exchange or conversion of any security or loan, held by such Security Holder). If any such Security Holder would thus be entitled to include more securities than such Security Holder requested to be registered, the excess shall be allocated among the other Security Holders pro rata in a manner similar to that described in the previous sentence.
- (iv) The Company may amend any registration statement to withdraw registration of the Holder's Registrable Shares if the Holder fails or refuses to cooperate in full and in a timely manner with all reasonable requests relating to such registration and the public offering generally made by the Company, the underwriters (if any), their respective counsel and the Company's auditors.
- 10.5 Expenses. Without regard to whether the registration statement relating to the proposed sale of the Registrable Shares is made effective or the proposed sale of such shares is carried out, the Company shall pay the fees and expenses in connection with any such registration including, without limitation, legal, accounting and printing fees and expenses in connection with such registration statements, the registration filing and examination fees paid under the Securities Act and state securities laws and the filing fees paid to the National Association of Securities Dealers, Inc. Notwithstanding the foregoing, the Holder shall be responsible for the payment of underwriting discounts and commissions, if any, and applicable transfer taxes relating to the Registrable Shares sold by the Holder and for the fees and charges of any attorneys or other advisers retained by the Holder.
- 10.6 Indemnification. In the event any Registrable Shares are included in a registration statement under this Section 10:
- (a) To the extent permitted by law, with respect to each registration, qualification, or compliance that has been

effected pursuant to this Warrant, the Company will indemnify and hold harmless the Holder, his legal counsel and accountants (each a "Representative"), and any underwriter (as defined in the Securities Act) for the Holder and any controlling Person of such underwriter against any losses, claims, damages, or liabilities (joint or several) to which they may become subject under the Securities Act, the Exchange Act or other federal or state law, insofar as such expenses, losses, claims, damages, or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement, including any preliminary prospectus or final prospectus contained therein, offering circular or other document or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required or allegedly required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any other federal or state securities law or any rule or regulation promulgated under the Securities Act, the Exchange Act or any other federal or state securities law; and the Company will pay the Holder, the Holder's Representative, underwriter and any controlling Person of such underwriter or controlling Person any legal or other expenses reasonably incurred by such Person in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, or action if such settlement is effected without the consent of the Company (which consent shall not be unreasonably withheld), nor shall the Company be liable in any such case for any such loss, claim, damage, liability, or action to the extent that it arises out of or is based upon a Violation that occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by the Holder.

(b) To the extent permitted by law, the Holder will indemnify and hold harmless the Company, each of the Company's directors, each of the Company's officers who has signed the registration statement, each Person, if any, who controls the Company within the meaning of the Securities Act, any underwriter, any other Security Holder selling securities in such registration statement and any controlling Person of any such underwriter or other Security Holder, against any losses, claims, damages, or liabilities (joint or several) to which any of the foregoing Persons may become subject, under the Securities Act, the Exchange Act or other federal or state law, insofar as such losses, claims, damages, or

liabilities (or actions in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by the Holder expressly for use in connection with such registration; and the Holder will pay any legal or other expenses reasonably incurred by any Person intended to be indemnified pursuant to this subsection, in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Holder, which consent shall not be unreasonably withheld; provided, that, in no event shall any indemnity under this subsection exceed the net proceeds after unreimbursed expenses and commissions from the offering received by the Holder.

- (c) Promptly after receipt by an indemnified party under this Section of notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section, deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense of such action, with counsel mutually satisfactory to the parties; however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one separate counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action, if prejudicial to its ability to defend such action, shall relieve such indemnifying party of its liability to the indemnified party under this Section 10.6 only to the extent that the indemnifying party has been injured by the delay. The omission so to deliver written notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section.
- (d) If the indemnification provided for in this Section is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage, or expense referred to therein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage, or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other in connection with the statements or omissions that resulted in such loss, liability, claim, damage, or expense as well as any other relevant equitable considerations. The relative

fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information, and opportunity to correct or prevent such statement or omission.

- (e) No indemnifying party, in defense of any such claim or litigation, shall, except with the consent of each indemnified party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnifying party of a release from all liability in respect to such claim or litigation.
- (f) To the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with any underwritten public offering are in conflict with the foregoing provisions, the provisions in this Warrant shall control.
- (g) The obligations of the Company and the Holder under this Section 10.6 shall survive the completion of any offering of Registrable Shares in a registration statement under this Section 10.6, and otherwise.
- 10.7 Information by the Holder. The Holder shall furnish to the Company such information regarding the Holder and the distribution proposed by him as the Company may reasonably request in writing and as shall reasonably be required in connection with any registration or qualification referred to in this Section 10.
- 10.8 SEC Rule 144 Reporting and Reports Under Securities Act and Exchange Act. With a view to making available to the Holder the benefits of SEC Rule 144 promulgated under the Securities Act and any other rule or regulation of the SEC that may at any time permit the Holder to sell securities of the Company to the public without registration or pursuant to a registration on Form S-3 or its successor, the Company agrees to:
- (a) make and keep public information available, as those terms are understood and defined in Rule 144, at all times from and after ninety (90) days following the effective date of the first registration statement filed by the Company for the offering of its securities to the general public;
- (b) take such action, including the voluntary registration of its Common Stock under Section 12 of the Exchange Act, as is necessary to enable the Holder to utilize Form S-3 or its successor for the sale of his Registrable Shares, such action to be taken as soon as practicable after the end of the fiscal year in which the first registration statement filed by the Company for the offering

of its securities to the general public is declared effective;

- (c) file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act after it has become subject to such reporting requirements; and
- furnish to the Holder, so long as the Holder owns any Registrable forthwith upon request (i) a written statement by the Company that it Shares, has complied with the reporting requirements of SEC Rule 144 (at any time from after ninety (90) days following the effective date of the first registration statement filed by the Company for an offering of the securities to the general public), the Securities Act and the Exchange Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 or its successor (at any time after it so qualifies), (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by Company (at any time after it has become subject to such reporting requirements), and (iii) such other information as may be reasonably requested in availing the Holder of any rule or regulation of the SEC which permits the selling of any such securities without registration or pursuant to such Form S-3 or its successor.
 - 10.9 Transfer or Assignment of Registration Rights. The rights to cause the Company to register Registrable Shares pursuant to this Section 10 may be transferred or assigned (but only with all related obligations) by the Holder to a transferee or assignee of such securities, provided: (a) such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Securities Act; and (b) such assignment shall only be effective if it complies with all applicable federal and state securities laws. For the purposes of determining the number of shares of Registrable Securities held by a transferee or assignee, the holdings of transferees and assignees of a partnership who are partners or retired partners of such partnership (including spouses and ancestors, lineal descendants and siblings of such partners or spouses who acquire Registrable Shares by gift, will or intestate succession) shall be aggregated together and with the partnership.
 - 10.10 Priority and Limitation on Subsequent Registration Rights.
- (a) The parties hereto acknowledge that the rights to registration contained herein shall be subject to (i) the registration rights contained in Section 2(k) of those certain Registration Rights Agreements ("Registration Rights Agreements") dated October 6, 1996 by and among InfoMed Holdings, Inc. (as predecessor in interest to the Company) and certain shareholders of the Company named therein, the registration rights granted pursuant to that certain

Second Amended and Restated Agreement and Plan of Merger and Investment Agreement dated as of October 25, 1999 among MCS, Inc., Mestek, Inc., the Company, John E. Reed ("Reed"), Stewart B. Reed and E. Herbert Burk (the "MCS Merger Agreement"), and (iii) the registration rights granted pursuant to that certain Agreement and Plan of Merger dated as of July 12, 1999 among CareCentric Solutions, Inc., Simione Acquisition Corporation and the Company (the "CareCentric Merger Agreement"): provided that the registration rights set forth in the Registration Rights Agreements, the MCS Merger Agreement and the CareCentric Merger Agreement shall only have priority over the registration rights granted pursuant to this Warrant to the extent required in such agreements and to the extent that any such prior rights have not been waived or amended.

- (b) Subject to Section 10.10(d), the Company will not grant any right of registration under the Securities Act relating to any of its equity securities to any person or entity other than pursuant to this Warrant unless the Holder shall be entitled to have included in such registration all Registrable Shares requested by the Holder to be so included prior to the inclusion of any securities requested to be registered by the persons or entities entitled to any such other registration rights, other than securities subject to the Registration Rights Agreements, the MCS Merger Agreement, and the CareCentric Merger Agreement, which shall have priority (but only to the extent that such prior rights have not been waived or amended).
- (c) Subject to Section 10.10(d), for so long as the Holder owns securities representing 20% or more of the voting power of the Company on a fully diluted basis, and except as expressly set forth in this Section 10.10, no other Person shall be entitled to "piggyback" or participate in any of the demand registrations that the Holder initiates pursuant to Section 10.2 without such Holder's prior written consent.
- (d) The parties agree that the rights to registration contained herein shall be pari passu with the rights to registration granted in (i) that certain Amended and Restated Secured Convertible Credit Facility and Security Agreement between the Company and Reed, dated as of July 1, 2002; (ii) that certain Series D Convertible Preferred Stock Purchase Agreement between the Company and Reed, dated June 12, 2000, as amended or restated and (iii) any other warrants issued to the Holder whether now existing or hereafter issued.
- 10.11 Suspension of Registration Rights. The right of the Holder to request registration of shares as provided in this Section 10 shall be suspended during any period of time that all of the Registrable Shares held and entitled to be held (as a result of conversion of Series D Preferred Stock held) by the Holder may immediately be sold under SEC Rule 144.

11. Miscellaneous.

- (a) Successors. All the covenants and provisions hereof by or for the benefit of the Company or the Holder shall bind and inure to the benefit of their respective successors and assigns.
- (b) Governing Law. This Warrant shall be deemed to be a contract made under the laws of the State of Delaware (notwithstanding any principles of conflicts of laws) and for all purposes shall be construed in accordance with the laws of said State.
- (c) Attorneys Fees in the Event of a Dispute. In the event of any action at law, suit in equity or arbitration proceeding in relation to this Warrant or any common stock issued or to be issued hereunder, the prevailing party or parties shall be paid by the other party or parties a reasonable sum for attorneys, fees and expenses of such prevailing party or parties.
- (d) Saturdays. Sundays. Holidays. If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall be a Saturday or a Sunday or shall be a legal holiday in the State of Delaware, then such action may be taken or such right may be exercised on the next succeeding day not a legal holiday.
- (e) Amendment. This Warrant and any term hereof may not be changed, waived, discharged or amended except by an instrument in writing signed by the party against whom enforcement of such change, waiver, discharge or amendment is sought.
- (f) Multiple Counterparts. This Warrant may be executed in multiple counterparts, each of which shall for all purposes be deemed to be an original and all of which shall constitute the same instrument.

IN WITNESS WHEREOF, the Company has caused this Warrant to be executed by its duly authorized officer as of the 1st day of July, 2002.

	CareCentric, Inc.
	By_/S/ JOHN FESTA
HOLDER: Mestek, Inc.	Its: PRESIDENT
Ву	
Its:	

WARRANT

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 OR ANY STATE SECURITIES LAWS AND, UNLESS SO REGISTERED, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND APPLICABLE STATE SECURITIES LAWS.

Warrant to purchase 400,000 shares of the \$0.001 par value common stock of CareCentric, Inc. (subject to adjustment)

WARRANT TO PURCHASE COMMON STOCK OF CARECENTRIC, INC.

This certifies that, for value received, Mestek, Inc., or its successors or assigns (the "Holder"), is entitled, subject to the terms set forth below, to purchase CareCentric, Inc. (the "Company") up to 400,000 shares of the \$0.001 par value common stock of the Company ("Common Stock"), as the Company is constituted on the 1st day of July, 2002 (the "Warrant Issue Date"), upon surrender of this certificate at 2625 Cumberland Parkway, Suite 310, Atlanta Georgia, or such other place as the Company may designate in writing to the Holder, and the simultaneous payment therefor in lawful money of the United States of America of the Exercise Price (as hereinafter defined). The number, character and Exercise Price of such shares are subject to adjustment as provided herein. The term "Warrant" as used herein shall include this certificate, the securities represented by this certificate and any warrants delivered in substitution or exchange for this certificate as provided herein.

- 1. Term of Warrant. Subject to the terms and conditions set forth herein, this Warrant shall be exercisable, in whole or in part, during the period of time (the "Exercise Period") commencing on the Warrant Issue Date and ending at 5:00 p.m. on June 15, 2004, and shall be void thereafter.
- 2. Exercise Price. The price at which the Holder may exercise this Warrant (the "Exercise Price") shall be \$1.00 per share, subject, however, to adjustments as provided in Section 9 hereof.
- 3. Vesting of Warrant. Effective as of the Warrant Issue Date, the Warrant shall be fully vested and exercisable, and the Holder shall have the fully vested right to purchase 400,000 shares of Common Stock pursuant to the terms and conditions of this Warrant.
- 4. Exercise of the Warrant. The purchase rights represented by this Warrant

are exercisable by the Holder, in whole or in part, at any time, and from time to time during the Exercise Period, by the Holder's surrender of this Warrant at 2625 Cumberland Parkway, Suite 310, Atlanta Georgia, or such other place as the Company may designate in writing to the Holder, and the simultaneous payment therefor in lawful money of the United States of America of the Exercise Price in immediately available funds. This Warrant shall be deemed exercised on the date immediately prior thereto, and the Holder shall be entitled to receive the shares of Common Stock and be treated for all purposes as the holder of record of such shares as of the close of business on such date. As promptly as practicable, but in no event later than ten (10) business days thereafter, the Company shall issue and deliver, at its sole cost and expense, to the person or persons entitled to receive the same a certificate or certificates for the

number of shares issuable upon such exercise. In the event that this Warrant is exercised in part, the Company, at its sole cost and expense, shall execute and deliver a new warrant of like tenor as this Warrant, exercisable for the remaining number of shares for which this Warrant may then be exercised, and shall cancel this Warrant only upon issuance of such new warrant. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant, and in lieu thereof, the Company shall make a cash payment to the Holder equal to the Exercise Price multiplied by such fraction.

- 5. Rights as a Stockholder. The Holder shall not be entitled to vote, receive dividends or be deemed to be the owner of record of the shares of Common Stock to which this Warrant relates unless and until the Holder exercises this Warrant, and then the Holder shall enjoy such rights only to the extent of such exercise.
- 6. Transfer of Warrant.
- (a) Warrant Register. The Company will maintain a register (the "Warrant Register") maintaining the names and addresses of the Holder or Holders. Any Holder of this Warrant or any portion thereof may change its address as shown on the Warrant Register by written notice to the Company requesting such change. Any notice or written communication required or permitted to be given to the holder may be delivered or given by mail to such Holder as shown on the Warrant Register and at the address shown on the Warrant Register. Until this Warrant is transferred on the Warrant Register of the Company, the Company may treat the Holder as shown on the Warrant Register as the absolute owner of this Warrant for all purposes, notwithstanding any notice to the contrary.
- (b) Warrant Agent. The Company may, by written notice to the Holder, appoint an agent for the purpose of maintaining the Warrant Register referred to

in Section 6(a) above, issuing the Common Stock or other securities then issuable upon the exercise of this Warrant, exchanging this Warrant, replacing this Warrant, or any or all of the foregoing. Thereafter, any such registration, issuance, exchange, or replacement, as the case may be, shall be made at the office of such agent.

- (c) Transferability of Warrant. This Warrant may not be transferred or assigned (i) except in its entirety (other than transfers to subsidiaries or affiliates of Mestek, Inc.) and (ii) without compliance with all applicable federal and state securities laws by the transferor and the transferee (including delivery of investment representation letters reasonably satisfactory to the Company, if such are requested by the Company), and then only against receipt of an agreement of the transferee to comply with the provisions of this Section 6(c) with respect to any resale or other disposition of this Warrant.
- (d) Exchange of Warrant upon a Transfer. On surrender of this Warrant for exchange, properly endorsed and subject to the provisions of this Warrant with respect to compliance with the Act and with the limitations on assignments and transfers contained in this Section 6, the Company at its expense shall issue to or on the order of the Holder a new Warrant or Warrants of like tenor, in the name of the Holder or as the Holder (on payment by the Holder of any applicable transfer taxes) may direct, for the number of shares issuable upon exercise hereof.
- (e) Compliance with Securities Laws.
- (i) The Holder of this Warrant, by acceptance hereof, acknowledges that this Warrant and the shares of Common Stock to be issued upon exercise hereof are being acquired solely for the Holder's own account and not as a nominee for any other party, and for investment, and that the Holder will not offer, sell or otherwise dispose of this Warrant or any shares of the Common Stock to be issued upon exercise hereof except under circumstances that will not result in a violation of the Securities Act of 1933 or any state securities laws. Upon exercise of this Warrant, the Holder shall, if requested by the Company, confirm in writing, in a form satisfactory to the Company, that the shares of Common Stock so purchased are being acquired solely for the Holder's own account and not as a nominee for any other party, for investment, and not with a view toward distribution or resale.
- (ii) This Warrant and all shares of Common Stock issued upon exercise hereof or conversion thereof shall be stamped or imprinted with a legend in substantially the following form (in addition to any legend required by state securities laws): THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN

REGISTERED UNDER THE SECURITIES ACT OF 1933 OR ANY STATE SECURITIES LAWS AND, UNLESS SO REGISTERED, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND APPLICABLE STATE SECURITIES LAWS.

- Reservation of Stock. The Company covenants that during the Exercise Period, the Company will reserve from its authorized and unissued shares of treasury Common Stock a sufficient number of shares to provide for the issuance of Common Stock upon the exercise of the Warrant and, from time to time, will take all steps necessary to amend its certificate of incorporation (the "Certificate") to provide sufficient authorized reserved shares of Common Stock issuable upon exercise of the Warrant. The Company further covenants that all shares that may be issued upon exercise of the rights represented by this Warrant and payment of the Exercise Price, all as set forth herein, will be free from all taxes, liens and charges in respect of the issue hereof (other than taxes in respect of any transfer occurring contemporaneously or otherwise specified herein). The Company agrees that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of executing stock certificates to execute and issue the necessary certificates for shares of Common Stock upon the exercise of this Warrant.
- 8. Merger; Sale of Assets and other Fundamental Corporate Changes. If at any time during the Exercise Period there shall be a sale of all or substantially all of the Company assets, or a merger, consolidation or reorganization of the Company in which the Company is not the surviving entity, or other transaction in which the shares of the Company are converted into shares of another entity, the Company shall provide the Holder with written notice thereof not less than 30 calendar days prior to the consummation of such event and an opportunity to exercise this Warrant prior to the consummation of such event.
- 9. Adjustments to Exercise Price for Certain Diluting Issuances. Splits and Combinations. The Exercise Price shall be subject to adjustment from time to time as follows:
- (a) Special Definitions. For purposes of this Section 9, the following definitions apply:
- (i) "Options" shall mean rights, options, or warrants to subscribe for, purchase or otherwise acquire either Common Stock or Convertible Securities (defined below).

"Convertible Securities" shall mean any evidences of indebtedness, shares (other than Common Stock) or other securities convertible into or exchangeable for Common Stock.

"Additional Stock" shall mean all shares of Common Stock issued by the Company after the Warrant Issue Date, and all shares of Common Stock issuable pursuant to Options and Convertible Securities issued by the Company after the Warrant Issue Date, other than shares of Common Stock for which adjustment of the Exercise Price is made pursuant to Section 9(d) or 9(e) below.

- (b) Adjustments. If the Company shall issue, after the Warrant Issue Date, any Additional Stock without consideration or for a consideration per share less than the Exercise Price in effect immediately prior to the issuance of such Additional Stock, the Exercise Price in effect immediately prior to each such issuance shall forthwith be adjusted downward to a price equal to the price paid per share for such Additional Stock.
- (c) Determination of Consideration. For purposes of this Section 9, the consideration received by the Company for the issuance of any Additional Stock shall be computed as follows:
 - (i) Cash and Property. Such consideration shall:
- (A) insofar as it consists of cash, be computed at the aggregate amount of cash received by the Company excluding amounts paid or payable for accrued interest or accrued dividends;
- (B) insofar as it consists of property other than cash, be computed at the fair value thereof at the time of such issue, as determined in good faith by the Company's Board of Directors; and
- (C) in the event Additional Stock is issued together with other shares or securities or other assets of the Company for consideration which covers both, be the proportion of such consideration so received, computed as provided above, as determined in good faith by the Company's Board of Directors.
- (ii) Options and Convertible Securities. The consideration per share received by the Company for Additional Stock deemed to have been issued pursuant to this Section 9 relating to Options and Convertible Securities shall be determined by dividing:
- (A) the total amount, if any, received or receivable by the Company as consideration for the issuance of such Options or Convertible Securities, plus the minimum aggregate amount of additional consideration (as set forth in the instruments relating thereto, without regard

to any provision contained therein designed to protect against dilution) payable to the Company upon the exercise of such Options or the conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities, by

- (as set forth in the instruments relating thereto, without regard to any provision contained therein designed to protect against dilution) issuable upon the exercise of such Options or conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities.
- Adjustments to Exercise Price for Stock Dividends and for Combinations or (d) Subdivisions of Common Stock. In the event that the Company at any time or from time to time after the Warrant Issue Date shall declare or pay any dividend on the Common Stock payable in Common Stock or in any right to acquire Common Stock for no consideration, or shall effect a subdivision of the outstanding shares of Common Stock into a greater number of shares of Common Stock (by stock split, reclassification or otherwise), or in the event the outstanding shares of Common Stock shall be combined or consolidated, by reclassification or otherwise, into a lesser number of shares of Common Stock, then the Exercise Price in effect immediately prior to such event shall, concurrently with the effectiveness of such event, be proportionately decreased or increased, as appropriate. In the event that the Company shall declare or pay any dividend on the Common Stock payable in any right to acquire Common Stock for no consideration, then the Company shall be deemed to have made a dividend payable in Common Stock in an amount of shares equal to the maximum number of shares issuable upon exercise of such rights to acquire Common Stock.
- (e) Adjustments for Reclassification and Reorganization. If the Common Stock shall be changed into the same or a different number of shares of any other class or classes of stock, whether by capital reorganization, reclassification or otherwise (other than a subdivision or combination of shares provided for in Section 9(d) above) the applicable Exercise Price then in effect shall, concurrently with the effectiveness of such reorganization or reclassification, be proportionately adjusted so that the shares of Common Stock issued upon exercise hereof shall be convertible into, in lieu of the number of shares of Common Stock which the Holder would otherwise have been entitled to receive, a number of shares of such other class or classes of stock equivalent to the number of shares of Common Stock that would have been subject to receipt by the Holder upon conversion of shares of the Common Stock immediately before that change.
- 10. Registration Rights. The Company covenants and agrees that the Holder shall have registration rights with respect to this Warrant as follows:

- 10.1 Definitions. For purposes of this Section 10:
- (a) The term "Form S-3" means such form under the Securities Act of 1933, as amended ("Securities Act"), as in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
- (b) The term "Form S-4" means such form under the Securities Act as in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC for corporate combinations and exchange offers which peanuts inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
- (c) The term "Holder" means the Holder or any permitted transferee or assignee thereof.
- (d) The term "Person" means an individual, a corporation, a limited liability company, a partnership, an association, a trust or any other entity organization, including a governmental entity.
- (e) This terms "register," "registered." and "registration" refer to a registration effected by preparing and filing a registration statement or similar document in compliance with the Securities Act, and the declaration or ordering of effectiveness of such registration statement or document.

The term "Registrable Shares" means (i) the Company's Common Stock issuable or issued upon exercise of this Warrant, and (ii) any Common Stock or other securities issued or issuable in respect of shares referenced in (i) above, upon any stock split, stock dividend, recapitalization, or similar event; excluding in all cases, however, any Registrable Shares sold by a Person in a transaction in which such Person's rights under this Section 10 are not assigned.

- (g) The term "SEC" means the Securities and Exchange Commission.
- (h) The term "Subsidiary" means, with respect to any Person, any corporation, limited liability company, or partnership of which such Person owns, either directly or through its subsidiaries or affiliates, more than fifty percent (50%) of (i) the total combined voting power of all classes of voting securities in the case of a corporation or (ii) the capital or profit

interests therein in the case of a partnership.

- Request for Registration. Upon request of the Holder, the Company will use its best efforts to file within 45 days of a request from the Holder a registration statement with the SEC (utilizing Form S-3 or a successor form thereto and Rule 415 to the extent available) to register Registrable Shares as requested by the Holder. The Company shall not be required to file more than three such registration statements (excluding any registration statement which is delayed pursuant to Section 10.4(e) below and through which the Holder is unable to register eighty percent (80%) or more of the amount of Registrable Shares that the Holder originally requested to register in such registration statement), and no such filing shall be made prior to the date which is six (6) months after the Warrant Issue Date.
- 10.3 Company Registration. If the Company at any time proposes to register an offering of its securities under the Securities Act, either for its own account or for the account of or at the request of one or more Persons holding securities of the Company, the Company will:
- (a) give written notice thereof to the Holder (which shall include a list of the jurisdictions in which the Company intends to attempt to qualify such securities under the applicable blue sky or other state securities laws) within 10 days of its receipt of a request from one or more Persons holding securities of the Company to register securities, or from its decision to effect a registration of securities for its own account, whichever first occurs; and
- (b) use its best efforts to include in such registration and in any underwriting involved therein, all the Registrable Shares specified in a written request by the Holder made within 30 days after receipt of such written notice from the Company, except as set forth in Section 10.4(e) below and subject to the currently existing piggyback rights referenced in Section 10.10.
- 10.4 Obligations of the Company. If and whenever pursuant to the provisions of this Section 10 the Company effects registration of Registrable Shares under the Securities Act and state securities laws, the Company shall:
- (a) Prepare and file with the SEC a registration statement with respect to such securities and use its best efforts to cause such registration statement to become and remain effective for a period not to exceed two years after the filing (but which period shall be extended by the duration of any delay periods under clause (e) below);
- (b) Use its best efforts to register or qualify the securities covered by such registration statement under the securities or blue sky laws of such jurisdictions as the Holder shall reasonably request, and do any and all other

acts and things which may be necessary or advisable (in the reasonable opinion of the Holder) to enable the Holder to consummate the disposition thereof; provided, however, that in no event shall the Company be obligated to qualify to do business in any jurisdiction where it is not now so qualified or to take any action which would subject it to the service of process in suits other than those arising out of the offer or sale of the securities covered by such registration statement in any jurisdictions where it is not now so subject;

- (c) As promptly as practicable prepare and file with the SEC such amendments and supplements to any registration statement and prospectus used pursuant to or in connection with this Agreement as may be necessary to keep such registration statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement or for such shorter period as may be required herein; and
- (d) Furnish to the Holder such number of conformed copies of its registration statement and of each such amendment and supplement thereto (in each case including all exhibits, such number of copies of the prospectus comprised in such registration statement (including each preliminary prospectus and any summary prospectus), in conformity with the requirements of the Securities Act), and such other related documents as the Holder may reasonably request in order to facilitate the disposition of the Registrable Shares to be registered.
- (e) Anything in this Agreement to the contrary notwithstanding:
 - (i) The Company may defer the filing ("Filing") of any registration statement or suspend the use of a prospectus under a currently effective registration statement under this Agreement at its discretion for "Good Cause." "Good Cause" means either if (1) the Company is engaged in active negotiations with respect to the acquisition of a "significant subsidiary" as defined in Regulation S-X promulgated by the SEC under the Securities Exchange Act of 1934, as amended ("Exchange Act") and the Securities Act which would in the opinion of counsel for the Company be required to be disclosed in the Filing; or (2) in the opinion of counsel for the Company, the Filing would require the inclusion therein of certified financial statements other than those in respect of the Company's most recently ended full fiscal year and any preceding full fiscal year, and the Company may then, at its option, delay the imposition of its registration obligations hereof until the earlier of (A) the conclusion or termination of such negotiations, or the date of availability of such certified financial statements, whichever is applicable, or

- (B) 60 days from the date of the registration request.
- (ii) In the event the Company has deferred a requested Filing, pursuant to the preceding paragraph, such deferral period shall end if the Company registers shares for resale by another stockholder of the Company. In the event the Company undertakes an underwritten public offering to issue the Company securities for cash during any period in which a requested Filing has been deferred or if the registration of which the Company gives notice under Section 10.3(a) is for an underwritten public offering to issue the Company securities for cash, the Company shall include the Registrable Shares in such underwritten offering subject to (A) the right of the managing underwriters to object to including such shares, (B) Section 10.10, and (C) the condition that the Holder selling Registrable Shares in such underwritten offering shall cooperate in the registration process in all material respects, including execution by the Holder of the underwriting agreement agreed to by the Company and the underwriters.
 - (iii) If the managing underwriter elects to limit the number or amount of securities to be included in any registration referenced in the preceding paragraph or in Section 10.3(a), all Persons holding securities of the Company (including the Holder) who hold registration rights and who have requested registration (collectively, the "Security Holders") shall, subject to Section 10.10 hereof, participate in the underwritten public offering pro rata based upon the ratio of the total number or amount of securities to be offered in the offering to the total number or amount of securities held by each Security Holder (including the number or amount of securities which each such Security Holder may then be entitled to receive upon the exercise of any option or warrant, or the exchange or conversion of any security or loan, held by such Security Holder). If any such Security Holder would thus be entitled to include more securities than such Security Holder requested to be registered, the excess shall be allocated among the other Security Holders pro rata in a manner similar to that described in the previous sentence.
- (iv) The Company may amend any registration statement to withdraw registration of the Holder's Registrable Shares if the Holder fails or refuses to cooperate in full and in a timely manner with all reasonable requests relating to such registration and the public offering generally made by the Company, the underwriters (if any), their respective counsel and the Company's auditors.
- 10.5 Expenses. Without regard to whether the registration statement relating to the proposed sale of the Registrable Shares is made effective or the proposed sale of such shares is carried out, the Company shall pay the fees and

expenses in connection with any such registration including, without limitation, legal, accounting and printing fees and expenses in connection with such registration statements, the registration filing and examination fees paid under the Securities Act and state securities laws and the filing fees paid to the National Association of Securities Dealers, Inc. Notwithstanding the foregoing, the Holder shall be responsible for the payment of underwriting discounts and commissions, if any, and applicable transfer taxes relating to the Registrable Shares sold by the Holder and for the fees and charges of any attorneys or other advisers retained by the Holder.

10.6 Indemnification. In the event any Registrable Shares are included in a registration statement under this Section 10:

(a)

To the extent permitted by law, with respect to each registration, qualification, or compliance that has been effected pursuant to this Warrant, the Company will indemnify and hold harmless the Holder, his legal counsel and accountants (each a "Representative"), and any underwriter (as defined in the Securities Act) for the Holder and any controlling Person of such underwriter against any losses, claims, damages, or liabilities (joint or several) to which they may become subject under the Securities Act, the Exchange Act or other federal or state law, insofar as such expenses, losses, claims, damages, or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement, including any preliminary prospectus or final prospectus contained therein, offering circular or other document or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required or allegedly required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any other federal or state securities law or any rule or regulation promulgated under the Securities Act, the Exchange Act or any other federal or state securities law; and the Company will pay the Holder, the Holder's Representative, underwriter and any controlling Person of such underwriter or controlling Person any legal or other expenses reasonably incurred by such Person in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, or action if such settlement is effected without the consent of the Company (which consent shall not be unreasonably withheld), nor shall the Company be liable in any such case for any such loss, claim, damage, liability, or action to the extent that it arises out of or is based upon a Violation that occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by the Holder.

- (b) To the extent permitted by law, the Holder will indemnify and hold harmless the Company, each of the Company's directors, each of the Company's officers who has signed the registration statement, each Person, if any, who controls the Company within the meaning of the Securities Act, any underwriter, any other Security Holder selling securities in such registration statement and any controlling Person of any such underwriter or other Security Holder, against any losses, claims, damages, or liabilities (joint or several) to which any of the foregoing Persons may become subject, under the Securities Act, the Exchange Act or other federal or state law, insofar as such losses, claims, liabilities (or actions in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by the Holder expressly for use in connection with such registration; and the Holder will pay any legal or other expenses reasonably incurred by any Person intended to be indemnified pursuant to this subsection, in connection with investigating or defending any such loss, claim, damage, liability, action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Holder, which consent shall not be unreasonably withheld; provided, that, in no event shall any indemnity under this subsection exceed the net proceeds after unreimbursed expenses and commissions from the offering received by the Holder.
- (c) Promptly after receipt by an indemnified party under this Section of notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section, deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense of such action, with counsel mutually satisfactory to the parties; however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one separate counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action, if prejudicial to its ability to defend

such action, shall relieve such indemnifying party of its liability to the indemnified party under this Section 10.6 only to the extent that the indemnifying party has been injured by the delay. The omission so to deliver written notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section.

- (d) If the indemnification provided for in this Section is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage, or expense referred to therein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage, or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other in connection with the statements or omissions that resulted in such loss, liability, claim, damage, or expense as well as any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information, and opportunity to correct or prevent such statement or omission.
- (e) No indemnifying party, in defense of any such claim or litigation, shall, except with the consent of each indemnified party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnifying party of a release from all liability in respect to such claim or litigation.
- (f) To the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with any underwritten public offering are in conflict with the foregoing provisions, the provisions in this Warrant shall control.
- (g) The obligations of the Company and the Holder under this Section 10.6 shall survive the completion of any offering of Registrable Shares in a registration statement under this Section 10.6, and otherwise.
- 10.7 Information by the Holder. The Holder shall furnish to the Company such information regarding the Holder and the distribution proposed by him as the Company may reasonably request in writing and as shall reasonably be required in connection with any registration or qualification referred to in this Section 10.
- 10.8 SEC Rule 144 Reporting and Reports Under Securities Act and Exchange Act. With a view to making available to the Holder the benefits of SEC Rule 144 promulgated under the Securities Act and any other rule or regulation of the SEC that may at any time permit the Holder to sell securities of the Company to the

public without registration or pursuant to a registration on Form S-3 or its successor, the Company agrees to:

- (a) make and keep public information available, as those terms are understood and defined in Rule 144, at all times from and after ninety (90) days following the effective date of the first registration statement filed by the Company for the offering of its securities to the general public;
- (b) take such action, including the voluntary registration of its Common Stock under Section 12 of the Exchange Act, as is necessary to enable the Holder to utilize Form S-3 or its successor for the sale of his Registrable Shares, such action to be taken as soon as practicable after the end of the fiscal year in which the first registration statement filed by the Company for the offering of its securities to the general public is declared effective;
- (c) file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act after it has become subject to such reporting requirements; and
- furnish to the Holder, so long as the Holder owns any Registrable (d) forthwith upon request (i) a written statement by the Company that it has complied with the reporting requirements of SEC Rule 144 (at any time from after ninety (90) days following the effective date of the first registration statement filed by the Company for an offering of the securities to the general public), the Securities Act and the Exchange Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 or its successor (at any time after it so qualifies), (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by Company (at any time after it has become subject to such reporting requirements), and (iii) such other information as may be reasonably requested in availing the Holder of any rule or regulation of the SEC which permits the selling of any such securities without registration or pursuant to such Form S-3 or its successor.
 - 10.9 Transfer or Assignment of Registration Rights. The rights to cause the Company to register Registrable Shares pursuant to this Section 10 may be transferred or assigned (but only with all related obligations) by the Holder to a transferee or assignee of such securities, provided: (a) such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Securities Act; and (b) such assignment shall only be effective if it complies with all applicable federal and state securities laws. For the purposes of determining the number of shares of Registrable Securities held by a transferee or assignee, the holdings of transferees and assignees of a partnership who are partners or retired partners of such partnership (including spouses and ancestors, lineal

descendants and siblings of such partners or spouses who acquire Registrable Shares by gift, will or intestate succession) shall be aggregated together and with the partnership.

- 10.10 Priority and Limitation on Subsequent Registration Rights.
- The parties hereto acknowledge that the rights to registration contained herein shall be subject to (i) the registration rights contained in Section 2(k) of those certain Registration Rights Agreements ("Registration Rights Agreements") dated October 6, 1996 by and among InfoMed Holdings, (as predecessor in interest to the Company) and certain shareholders of the Company named therein, the registration rights granted pursuant to that certain Second Amended and Restated Agreement and Plan of Merger and Investment Agreement dated as of October 25, 1999 among MCS, Inc., Mestek, Company, John E. Reed ("Reed"), Stewart B. Reed and E. Herbert Burk (the "MCS Merger Agreement"), and (iii) the registration rights granted pursuant to that certain Agreement and Plan of Merger dated as of July 12, 1999 among CareCentric Inc., Simione Acquisition Corporation and the Company (the "CareCentric Merger Agreement"): provided that the registration rights set forth in the Registration Rights Agreements, the MCS Merger Agreement and the CareCentric Merger Agreement shall only have priority over the registration rights granted pursuant to this Warrant to the extent required in such agreements and to the extent that any such prior rights have not been waived or amended.
- (b) Subject to Section 10.10(d), the Company will not grant any right of registration under the Securities Act relating to any of its equity securities to any person or entity other than pursuant to this Warrant unless the Holder shall be entitled to have included in such registration all Registrable Shares requested by the Holder to be so included prior to the inclusion of any securities requested to be registered by the persons or entities entitled to any such other registration rights, other than securities subject to the Registration Rights Agreements, the MCS Merger Agreement, and the CareCentric Merger Agreement, which shall have priority (but only to the extent that such prior rights have not been waived or amended).
- (c) Subject to Section 10.10(d), for so long as the Holder owns securities representing 20% or more of the voting power of the Company on a fully diluted basis, and except as expressly set forth in this Section 10.10, no other Person shall be entitled to "piggyback" or participate in any of the demand registrations that the Holder initiates pursuant to Section 10.2 without such Holder's prior written consent.
- (d) The parties agree that the rights to registration contained herein shall be pari passu with the rights to registration granted in (i) that certain Amended and Restated Secured Convertible Credit Facility and Security Agreement between the Company and Reed, dated as of July 1, 2002; (ii) that certain Series

D Convertible Preferred Stock Purchase Agreement between the Company and Reed, dated June 12, 2000, as amended or restated and (iii) any other warrants issued to the Holder whether now existing or hereafter issued.

10.11 Suspension of Registration Rights. The right of the Holder to request registration of shares as provided in this Section 10 shall be suspended during any period of time that all of the Registrable Shares held and entitled to be held (as a result of conversion of Series D Preferred Stock held) by the Holder may immediately be sold under SEC Rule 144.

11. Miscellaneous.

- (a) Successors. All the covenants and provisions hereof by or for the benefit of the Company or the Holder shall bind and inure to the benefit of their respective successors and assigns.
- (b) Governing Law. This Warrant shall be deemed to be a contract made under the laws of the State of Delaware (notwithstanding any principles of conflicts of laws) and for all purposes shall be construed in accordance with the laws of said State.
- (c) Attorneys Fees in the Event of a Dispute. In the event of any action at law, suit in equity or arbitration proceeding in relation to this Warrant or any common stock issued or to be issued hereunder, the prevailing party or parties shall be paid by the other party or parties a reasonable sum for attorneys, fees and expenses of such prevailing party or parties.
- (d) Saturdays. Sundays. Holidays. If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall be a Saturday or a Sunday or shall be a legal holiday in the State of Delaware, then such action may be taken or such right may be exercised on the next succeeding day not a legal holiday.
- (e) Amendment. This Warrant and any term hereof may not be changed, waived, discharged or amended except by an instrument in writing signed by the party against whom enforcement of such change, waiver, discharge or amendment is sought.
- (f) Multiple Counterparts. This Warrant may be executed in multiple counterparts, each of which shall for all purposes be deemed to be an original and all of which shall constitute the same instrument.

		CareCentric, Inc.			
		By_/S/ JOHN FESTA			
HOLDER: Mestek,	Inc.	Its: PRESIDENT			

IN WITNESS WHEREOF, the Company has caused this Warrant to be

executed by its duly authorized officer as of the 1st day of July, 2002.

AGREEMENT

This Agreement is made this 29th day of March, 2002, by and between Mestek, Inc., a Pennsylvania corporation ("Mestek") and John E. Reed, an individual residing in Massachusetts ("Reed").

- A. Each of Mestek and Reed have an investment in CareCentric, Inc., a Delaware corporation.
- B. Reed has provided a credit line to CareCentric that is subject to a declaration
- of default and is not currently eligible for further funding. Reed has also provided short-term working capital to CareCentric in the first quarter of 2002. Mestek has an interest in protecting its current investment by providing short-term working capital to CareCentric.
- C. CareCentric also obtained a credit
- line from Wainwright Bank, where Reed is a director, the payment of which is quaranteed by Mestek.
- D. CareCentric has requested that its short-term obligations to Mestek and Reed be refinanced to long-term financing. Mestek is willing to make such a commitment to CareCentric in consideration of various terms and conditions to re-capitalize and refinance its existing investment and certain agreements of Reed (the "Transaction").

NOW, THEREFORE, the parties hereto, intending to be legally bound, for good and adequate consideration to hereby agree as follows:

1. Reed Forebearance.

Reed hereby agrees, pending the period commencing the date hereof to the closing of the Transaction, to forbear from declaring a default under any of CareCentric's obligations to him and to waive any default thereunder. In addition, Reed agrees that in any future actions to pursue remedies regarding any CareCentric obligations to him, that he will give Mestek prior written notice and seek to coordinate his actions with Mestek.

- 2. Other Reed Agreements.
- Reed further agrees to use his good offices to advocate for the continued extension of credit by Wainwright Bank to CareCentric as warranted by the financial condition of CareCentric.
- 3. Reed Committment. Reed further agrees to submit to CareCentric, simultaneously
- with the submission of any similar commitment of Mestek, but not later than April 12, 2002, a written and binding commitment (the "Reed Commitment Letter") for the Transaction, including a credit facility for CareCentric of approximately \$900,000 of new money, on terms that are substantially similar to those offered by Mestek and are otherwise acceptable to Mestek, including substantially similar conditions and contingencies to the timing and enforceability of the Reed Commitment Letter, the restructuring of existing debt, the recapitalization of the Series D Preferred Stock of CareCentric and the pay-down of CareCentric's Wainwright facility.
- 4. Transfer of Voting and Other Rights. In consideration of the foregoing

agreements, any one of which would be adequate, and the further conditions set forth in this section, Mestek hereby agrees to transfer to Reed, effective the date set forth above, all right, title and interest it has or may have in, to and under all votes it has or may have in all matters to be voted upon by the stockholders of CareCentric and any right it has to otherwise affect or influence the appointment of members of or nominees to the Board of Directors of CareCentric under the Certificate of Designations, Preferences and Rights of Series B Preferred Stock of CareCentric, as amended, or otherwise, so long as Mestek shall have any rights under the Series B Preferred Stock Certificate.

5. Effect of Transfer. This Agreement has no effect on the rights of Mestek under

the other terms of the Certificate of Designations, Preferences and Rights of Series B Preferred Stock of CareCentric, as amended, not specifically transferred hereunder, nor under the Certificate of Designations, Preferences and Rights of Series C Preferred Stock of CareCentric, as amended, nor any other investment of Mestek in CareCentric.

6. Notice to CareCentric. Mestek shall give CareCentric written notice of this Agreement transferring the Series B Preferred Stock votes prior to the next record date.

MESTEK. INC.

By:_/S/ R.BRUCE	DEWEY	/S/	JOHN	Ε.	REED
			JOHN	Ε.	REED

Its: PRESIDENT AND COO

SECURED CONVERTIBLE CREDIT FACILITY AND SECURITY AGREEMENT

By and Between

CARECENTRIC, INC.,

SC HOLDING, INC.,

CARECENTRIC NATIONAL, LLC

and

MESTEK, INC.

Dated as of July 1, 2002

SECURED CONVERTIBLE CREDIT FACILITY AND SECURITY AGREEMENT

THIS SECURED CONVERTIBLE CREDIT FACILITY AND SECURITY AGREEMENT (this "Agreement"), dated as of July 1, 2002, is made and entered into by and between CareCentric, Inc. ("CareCentric"), SC Holding, Inc. ("SCH"), and CareCentric National, LLC ("CCN") (CareCentric, SCH, and CCN, collectively, referred to hereinafter as "Borrower"), and Mestek, Inc. ("Lender").

WITNESSETH:

WHEREAS, Lender has agreed to provide a secured convertible term loan facility for five years in a principal amount of four million dollars (\$4,000,000) (the "Facility") into Borrower to refinance existing debt and fund working capital needs of Borrower; and

WHEREAS, Lender shall have the option at any time prior to payment in full to convert the Facility (including accrued but unpaid interest), in whole or in part, to common stock of CareCentric;

NOW, THEREFORE, in consideration of the premises and mutual agreements herein contained, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1 Definitions

In addition to any terms defined elsewhere in this Agreement, the following terms have the meanings indicated for purposes of this Agreement (such definitions being equally applicable to the singular and plural forms of the defined term):

"Acceleration" means that the Loan (i) shall not have been paid at the Maturity Date, or (ii) shall have become due and payable prior to the Maturity Date pursuant to Section 7.2.

"Account" means all "accounts" (as defined in the UCC) now owned or hereafter created or acquired by Borrower including, without limitation, all of the following now owned or hereafter created or acquired by Borrower: (a) accounts receivable, contract rights, book debts, notes, drafts and other obligations or indebtedness owing to Borrower arising from the sale, lease or exchange of goods or other property and/or the performance of services; (b) Borrower's rights in, to and under all purchase orders for goods, services or other property; (c) Borrower's rights to any goods, services or other property represented by any of the foregoing (including returned or repossessed goods and unpaid seller's rights of rescission, replevin, reclamation and rights to stoppage in transit); (d) monies due to or to become due to Borrower under all contracts for the sale, lease or exchange of goods or other property and/or the performance of Borrower); and (e) Proceeds of any of the foregoing and all collateral security and guaranties of any kind given by any person with respect to any of the foregoing.

"Accountants" shall mean the Borrower's independent accounting firm selected by the Borrower's board of directors.

"Affiliate" means, with respect to any Person, any Person directly or indirectly controlling, controlled by or under common control with such Person; provided, however, that neither party to this Agreement shall be deemed to be an Affiliate of the other party.

"Agreement" or "Loan Agreement" means this Secured Convertible Credit Facility and Security Agreement, as amended from time to time.

"Assigned Agreements" means all significant agreements of Borrower.

"Bank" means the Wainwright Bank & Trust Company, Boston, Massachusetts.

"Business" shall refer to the healthcare information systems and

consulting services operations of Borrower.

"Business Day" means a day when Lender and Borrower are both open for business.

"Chattel Paper" means a writing or writings which evidence both a monetary obligation and a security interest in or a lease of specific goods.

"Closing Date" means the effective date of this Agreement.

"Collateral" refers to the following: (i) all of Borrower's Inventory, Equipment and Fixtures now owned or hereafter acquired; (ii) all of Borrower's Documents of Title now owned or hereafter acquired; (iii) all of Borrower's Accounts now existing or hereafter arising; (iv) all of Borrower's Contract Rights now existing or hereafter arising; (v) all of Borrower's General Intangibles, Chattel Paper and Instruments, now existing or hereafter acquired or arising; (vi) all suretyships and guarantees of Borrower's existing and future Accounts, Contract Rights and General Intangibles and all security for the payment or satisfaction of such suretyships and guarantees; (vii) the goods or the services the sale or lease or performance of which gave rise to any Account, Contract Right or General Intangible of Borrower including any returned goods; (viii) any balance or share belonging to Borrower of any deposit, agency or other account with any bank and any other amounts which may be owing from time to time by any bank to Borrower; (ix) all property of any nature whatsoever of Borrower now or hereafter in the possession of or assigned or hypothecated to the Lender for any purpose; and (x) all Products and Proceeds of all of the foregoing, including all Proceeds of other Proceeds.

"Common Stock" means the common stock, par value of \$.001 per share, of CareCentric.

"Contract Right" means any right to payment under a contract (including, but not limited to, contracts for the sale or leasing of goods or for the rendering of services) not yet earned by performance and not evidenced by an Instrument or Chattel Paper.

"Contingent Obligation", as applied to any Person, means any direct or indirect liability, contingent or otherwise, of that Person: (a) with respect to any indebtedness, lease, dividend or other obligation of another Person if the primary purpose or intent of the Person incurring such liability, or the primary effect thereof, is to provide assurance to the obligee of such liability that such liability will be paid or discharged, or that any agreements relating hereto will be complied with, or that the holders of such liability will be protected (in whole or in part) against loss with respect thereto; (b) with respect to any letter of credit issued for the account of that Person or as to which that Person is otherwise liable for reimbursement of drawings; or (c) under any foreign exchange contract, currency swap agreement, interest rate swap or cap agreement or other similar agreement or arrangement designed to protect that Person against fluctuations in currency values or interest rates. Contingent Obligations shall include, without limitation, (i) the direct or indirect guaranty, endorsement (otherwise than for collection or deposit in the

ordinary course of business), co-making, discounting with recourse or sale with recourse by such person of the obligation of another, (ii) the obligation to make take-or-pay or similar payments if required regardless of nonperformance by any other party or parties to an agreement, and (iii) any liability of such Person for the obligations of another through any agreement to purchase, repurchase or otherwise acquire such obligation, to provide funds for the payment or discharge of such obligation or to maintain the solvency, financial condition or any balance sheet item or level of income of another. The amount of any Contingent Obligation shall be equal to the amount of the obligation so guarantied or otherwise supported or, if a fixed and determined amount, the maximum amount so guarantied.

"Copyright License" means any written agreement now or hereafter in existence granting to Borrower any right to use any Copyright.

"Copyrights" means collectively all of the following: (a) all copyrights, rights and interests in copyrights, copyright registrations and copyright applications now owned or hereafter created or acquired by Borrower; (b) all renewals of any of the foregoing; (c) all income, royalties, damages and payments now or hereafter due and/or payable under any of the foregoing, including, without limitation, damages or payments for past or future infringements of any of the foregoing; (d) the right to sue for past, present and future infringements of any of the foregoing; and (e) all rights corresponding to any of the foregoing throughout the world.

"Document of Title" means a bill of lading, dock warrant, dock receipt, warehouse receipt or order for the delivery of goods, and also any other document which in the regular course of business or financing is treated as adequately evidencing that the Person in possession of it is entitled to receive, hold and dispose of the document and the goods it covers.

"Equipment" means all "equipment" (as defined in the UCC) now owned or hereafter acquired by Borrower including, without limitation, all machinery, computers, computer equipment, motor vehicles, trucks, trailers, vessels, aircraft and rolling stock and all parts thereof and all additions and accessions thereto and replacements therefor.

"Event of Default" shall have the meaning set forth in Article 7 hereof.

"Financing Statements" shall mean the form of financing statements as shall be necessary to perfect, upon filing, a security interest in the Collateral in each jurisdiction in which such Collateral is located or in which a filing is required under the UCC to perfect such security interest.

"Fixtures" means all of the following now owned or hereafter acquired by Borrower: plant fixtures; business fixtures; other fixtures and storage office facilities, wherever located; and all additions and accessions thereto and replacements therefor.

"GAAP" means generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other entity as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" means all "general intangibles" (as defined in the UCC) now owned or hereafter acquired by Borrower including, without limitation, all right, title and interest of Borrower in and to: (a) the Assigned Agreements and all other agreements, leases, licenses and contracts to which Borrower is or may become a party; (b) all obligations or indebtedness owing to Borrower (other than accounts) from whatever source arising; (c) all tax refunds; (d) all Intellectual Property; and (e) all trade secrets, know-how and common law intellectual property rights including, without limitation, rights in computer software developed by or for Borrower and other confidential information relating to the business of the Borrower including by way of illustration and not limitation: the names and addresses of, and credit and other business information concerning, Borrower's past, present or future customers; the prices that Borrower obtains for its services or at which it sells merchandise; estimating and costs procedures; profit margins; policies and procedures pertaining to the sale and design of equipment, components, devices and services furnished by Borrower; information concerning suppliers of Borrower and information concerning the manner of operation, business plans, pledges, projections, and all other information of any kind or character, whether or not reduced to writing, with respect to the conduct by Borrower of its business not generally known by the public.

"Governmental Authority" means any nation or government, any state, province or other political subdivision thereof or any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Governmental Requirements" means all legal requirements in effect from time to time including all laws, statutes, codes, acts, ordinances, orders, judgments, decrees, injunctions, rules, regulations, permits, licenses, authorizations, certificates, orders, franchises, determinations, approvals, consents, notices, demand letters, directions and requirements of all governments, departments, commissions, boards, courts, authorities, agencies, officials and officers, and all instruments of record, foreseen or unforeseen, ordinary or extraordinary, including, but not limited to, any change in any law, regulation or the interpretation thereof by any foreign or domestic governmental or other authority (whether or not having the force of law), relating now or at any time heretofore or hereafter to the business or operations of Borrower or to any of the property owned, leased or used by Borrower, including, without

limitation, the development, design, construction, acquisition, start-up, ownership and operation and maintenance of property.

"Incipient Default" shall have the meaning set forth in Section 3.1(e).

"Indebtedness" of any Person means all liabilities, obligations and reserves, contingent or otherwise of such Person.

"Instrument" means a negotiable instrument or a security or any other writing which evidences a right to the payment of money and is not itself a security agreement or lease and is of a type which is in ordinary course of business transferred by delivery with any necessary indorsement or assignment.

"Intellectual Property" shall mean collectively all of the following: Copyrights, Copyright Licenses, Patents, Patent Licenses, Trademarks and Trademark Licenses.

"Inventory" means all "goods" (as defined in the UCC) now owned or hereafter acquired and held by Borrower for sale or lease or to be furnished under contracts of service, tangible personal property which Borrower has so leased or furnished, including, without limitation, tangible personal property, raw materials, work in process and materials used, produced or consumed in Borrower's business, and shall include tangible personal property held by others for sale on consignment from Borrower, tangible personal property sold by Borrower on a sale or return basis, tangible personal property returned to Borrower by the purchaser following a sale thereof by Borrower and tangible personal property represented by Documents of Title. All equipment, accessories and parts at any time attached or added to items of Inventory or used in connection therewith shall be deemed to be part of the Inventory.

"Liens" shall mean any lien, mortgage, pledge, security interest charge or encumbrance of any kind (including any conditional sale or other title retention agreement, any lease in the nature thereof and any agreement to give any security interest).

"Loan" shall have the meaning set forth in Section 2.1 hereof.

"Material Adverse Effect" means a material adverse effect on (i) the business, assets, operations or financial condition of Borrower, (ii) the ability of Borrower to pay the Obligations in accordance with their terms, or (iii) Lender's perfected first priority lien on the Collateral (as defined in this Agreement) or the value of such Collateral.

"Maturity" means any date on which the Loan or any portion thereof becomes due and payable, whether as stated or by virtue of mandatory prepayment, by Acceleration or otherwise.

"Maturity Date" means June 30, 2007.

"Note" means the promissory note executed by Borrower in the form of

"Obligations" means all loans, advances, debts, liabilities, obligations, covenants and duties owing to Lender by Borrower, of any kind or nature, present or future, whether or not evidenced by any note, guaranty or other instrument, arising under this Agreement or the Note, the Wainwright Guaranty, and all extensions, amendments, modifications, restructurings and refinancings of any of the above.

"O'Donnell Facility" means a \$600,000 credit facility provided by Barrett C. O'Donnell to the Borrower on August 8, 2000.

"Patent License" means any written agreement now or hereafter in existence granting to Borrower any right to use any invention on which a subsisting Patent exists.

"Patents" means collectively all of the following: (a) all patents and patent applications now owned or hereafter created or acquired by Borrower and the inventions and improvements described and claimed therein, and patentable inventions; (b) the reissues, divisions, continuations, renewals, extensions and continuations—in—part of any of the foregoing; (c) all income, royalties, damages or payments now and hereafter due and/or payable under any of the foregoing or with respect to any of the foregoing, including, without limitation, damages or payments for past of future infringements of any of the foregoing; (d) the right to sue for past, present and future infringements of any of the foregoing; and (e) all rights corresponding to any of the foregoing throughout the world.

"Person" means any individual, corporation, partnership, trust, association or other entity or organization, including any government, political subdivision, agency or instrumentality thereof.

"Proceeds" means all proceeds of, and all other profits, rentals or receipts, in whatever form, arising from the collection, sale, lease, exchange, assignment, licensing or other disposition of, or realization upon, any Collateral including, without limitation, all claims of Borrower against third parties for loss of, damage to or destruction of, or for proceeds payable for loss of, damage to or destruction of, or for proceeds payable under, or unearned premiums with respect to, policies of insurance with respect to any Collateral, and any condemnation or requisition payments with respect to any Collateral, in each case whether now existing or hereafter arising.

"Reed Facility" means a \$3,555,555 Secured Convertible Credit Facility (amended and restated as of the date of this Agreement) provided by John E. Reed to the Borrower.

"Restricted Payment" shall have the meaning set forth in Section 6.8 hereof.

"Trademark" means collectively all of the following now owned or hereafter created or acquired by Borrower: (a) all trademarks, trade names, corporate names, company names, business names, fictitious business names, trade styles, service marks, logos, other business identifiers, prints and labels on which any of the foregoing have appeared or appear, all registrations and recordings thereof, and all applications in connection therewith including registrations, recordings and applications in the United States Patent and Trademark Office or in any similar office or agency of the United States, any State thereof or any other country or any political subdivision thereof; (b) all reissues, extensions or renewals thereof; (c) all income, payable under any of the foregoing or with respect to any of the foregoing including damages or payments for past or future infringements of any of the foregoing; (d) the right to sue for past, present and future infringements of any of the foregoing; (e) all rights corresponding to any of the foregoing throughout the world; and (f) all goodwill associated with or symbolized by any of the foregoing.

"Trademark License" means any written agreement now or hereafter in existence granting to Borrower any right to use any Trademark.

"UCC" means the Uniform Commercial Code as in effect on the date hereof in the State of Delaware, as amended from time to time, and any successor statute; provided that if by reason of mandatory provisions of law, the perfection or the effect of perfection or non-perfection of the security interest in any Collateral is governed by the Uniform Commercial Code, or other applicable statute, law or provision relating to the perfection or the effect of perfection or non-perfection of any such security interest, as in effect on or after the date hereof in any other jurisdiction, "UCC" means the Uniform Commercial Code or such other statute, law or provision as in effect in such other jurisdiction for purposes of the provision hereof relating to such perfection or the effect of perfection or non-perfection.

"Wainwright Facility" means the \$6,000,000 Line of Credit facility provided by the Bank to the Borrower, or any modification, renewal, replacement or refinancing thereof.

"Wainwright Guaranty" means the guaranty provided to the Bank by the Lender, and any payments or performance made by the Lender on behalf of the Borrower thereunder.

Any accounting term not defined herein shall have the meaning given to it under GAAP.

ARTICLE 2 Term Loan

2.1 Term Loan . Subject to the terms and conditions of this Agreement and in reliance on the representations and warranties of Borrower set forth herein,

Lender agrees to make a convertible, secured term loan (the "Loan") to Borrower in the principal amount (excluding accrued interest added to principal as provided in 2.4(a)) of \$4,000,000. Borrower's obligation to repay the Loan shall be evidenced by a promissory note of Borrower (the "Note") in the form attached hereto as Exhibit A.

- 2.2 (Intentionally omitted)
- 2.3 Interest .
- (a) Interest. The Loan shall bear interest from the date of disbursement on the unpaid principal amount thereof until such amount is paid (whether upon Maturity, by Acceleration or otherwise) at a simple rate per annum equal to six and 25/100ths percent (6.25%), deferred until July 1, 2004 as set forth in Section 2.4(a) below.
- (b) Computation of Interest. Interest shall accrue daily and shall be computed for the actual number of days elapsed on the basis of a year consisting of 360 days.
- (c) Post-Maturity Interest. After Maturity (whether by acceleration or otherwise) of the Loan, the Loan shall bear interest, payable on demand, at a rate per annum equal to nine and one quarter percent (9.25%).
- (d) Maximum Interest Rate. Nothing in this Agreement shall require Borrower to pay interest at a rate exceeding the maximum amount permitted by applicable law to be charged by Lender (the "Maximum Rate"). If the amount of interest payable for the account of Lender on any day in respect of the immediately preceding interest computation period, computed pursuant to this Article 2, would exceed the Maximum Rate, the amount of interest payable for its account on such interest payment date shall automatically be reduced to the Maximum Rate.
 - 2.4 Payments .

(a) Payments of Loan. The payment of interest shall be deferred until June 30, 2004, at which time the accrued, but unpaid interest shall be capitalized and the face amount of the Note shall be deemed to have been increased to \$4,500,000. During the deferral period, interest shall not be compounded quarter to quarter (thus the total amount of interest to be capitalized and added to the principal as of June 30, 2004 shall be \$500,000). At the end of each calendar quarter thereafter (beginning on September 30, 2004) Borrower shall pay all interest accrued with respect to the preceding calendar quarter on the outstanding balance, including any capitalized interest. On the Maturity Date, Borrower shall pay (i) all accrued and unpaid interest on the Loan and (ii) the unpaid principal on the Loan.

- (b) Additional Payments. In addition to any payments under Section 2.4(a), Borrower shall immediately notify Lender of the occurrence of any of the following events and, if demanded by Lender, Borrower shall, within three (3) Business days of such demand make the following payments with respect to the Loan:
- (i) upon the sale, transfer, or other disposition of any asset of the Borrower or any of its subsidiaries with a value in excess of one million dollars (\$1,000,000) (other than the sale of inventory in the ordinary course of business), a payment in an amount equal to twenty-five percent (25%) of the net proceeds received from such sale, transfer, or other disposition;
- (ii) upon the sale of any equity securities issued by Borrower or any of its subsidiaries (other than equity securities issued to the Borrower or any of its subsidiaries or issued in the ordinary course of business pursuant to an employee benefit plan), a payment in an amount equal to twelve and one half percent (12.5%) of the net proceeds received in exchange for such equity securities; and
- (iii) upon issuance by Borrower or any of its subsidiaries of long-term debt securities (i.e., securities with a maturity date of one year or more from issuance) in the public or private capital markets, or incurrence by Borrower or any of its subsidiaries of indebtedness under one or more bank facilities (other than the Wainwright Facility, the Reed Facility and the O'Donnell Facility) in an aggregate principal amount in excess of one million dollars (\$1,000,000), a payment in an amount equal to twelve and one half percent (12.5%) of the principal amount.

Notwithstanding the foregoing requirements of this Section 2.4(b), Borrower shall not be required to make any payment under this Section 2.4(b) to the extent that such payment is prohibited by the terms of any debt of Borrower senior to the Loan, as listed on Schedule 6.1 hereto. To the extent that consent of the creditor with respect to any such senior debt is required in order to make any payment under this Section 2.4(b), Borrower shall make commercially reasonable efforts to obtain such consent.

- (c) Payments. All payments of interest and principal shall be in United States dollars and immediately available funds to Lender at its address for notices in this Agreement and shall be made prior to 1:00 P.M. Eastern Time on the date of the payment. All payments received after such time shall be credited the next succeeding Business Day, and interest shall continue to accrue.
- 2.5 Conversion Right . Lender may at any time at its option convert all or a portion of the unpaid interest and principal on the Loan into fully paid and nonassessable shares of Common Stock of CareCentric. The price at which shares of Common Stock shall be delivered to Lender (the "Conversion Price") shall be determined in accordance with Article 9 of this Agreement. The conversion of all

or any portion of a Loan shall, upon the issuance of certificates representing the shares of capital stock issuable upon such conversion, represent the full payment and satisfaction of that portion of the Loan which has been converted.

2.6 Subordination. The rights of the Lender to repayment of the Loan and the payment and performance of all Obligations hereunder shall be subordinate to the rights of the holders of instruments under the Reed Facility and the O'Donnell Facility.

ARTICLE 3 Conditions Precedent

- 3.1 Initial Loan . The obligation of Lender to make the Loan hereunder shall be subject to satisfaction or waiver of the following conditions precedent:
 - (a) Lender shall have received the Note, duly executed by Borrower;
 - (b) (Intentionally Omitted)
 - (c) Lender shall have received the Financing Statements, duly executed by Borrower;
 - (d) Lender shall have received such other instruments or documents as Lender may reasonably request relating to the existence and good standing of Borrower and the authority for execution, delivery and performance of this Agreement and the granting and/or perfection of security interest in the Collateral as contemplated herein;
 - (e) no Event of Default or event which, upon the lapse of time or the giving of notice or both, would constitute an Event of Default (an "Incipient Default") shall exist on the Borrowing Date;
 - (f) (Intentionally Omitted)
 - (g) the net worth (assets less liabilities, calculated in accordance with GAAP consistently applied) of CareCentric shall not have declined by more than twenty percent (20%) since December 31, 2001;
 - (h) (Intentionally Omitted);
 - (i) Lender shall have received an opinion of Borrower's

3.2 Subsequent Loan. The Lender has no obligation to make any subsequent Loan. If the Lender elects to do so, such subsequent Loan shall be subject to the terms, conditions and benefits of this Agreement and shall be subject to satisfaction or waiver of the same conditions precedent as are set forth in Section 3.1.

ARTICLE 4 [INTENTIONALLY OMITTED]

ARTICLE 5 Covenants

Unless Lender shall agree in writing otherwise, Borrower shall comply with the following provisions so long as any Loan is outstanding:

- 5.1 Accounting Records; Financial Statements . Borrower shall maintain adequate books and accounts in accordance with GAAP consistently applied. Borrower shall deliver to Lender any information regarding the Business or the finances of Borrower as Lender may reasonably request. On or before the ninetieth (90th) day after the end of Borrower's fiscal year, Borrower shall deliver to Lender annual audited financial statements, to include a balance sheet, an income statement, and a statement of cash flows. On or before the 45th day following the end of each fiscal quarter, Borrower shall provide Lender with the financial information filed with the Securities Exchange Commission on its Form 10-Q. On a monthly basis within five (5) business days of the end of each month Borrower shall provide Lender with a statement of cash flows, including a pro forma projection for 13 weeks following the date of the projection and actual cashflows for the 4 weeks prior thereto.
- 5.2 Corporate Existence . Borrower shall preserve and maintain its corporate existence in good standing in the jurisdiction of its formation and all of its licenses, privileges and franchises and other rights necessary or desirable in the ordinary course of its businesses, except to the extent that the failure to do so would not have a Material Adverse Effect.
- 5.3 Qualification to Do Business. Borrower shall qualify to do business and shall be and remain in good standing in each jurisdiction in which the nature of its business requires it to be so qualified, except to the extent that the failure to be so qualified and in good standing would not have a Material Adverse Effect.
- 5.4 Compliance with Laws . Borrower will observe and comply in all material respects with all laws, ordinances, orders, judgments, rules, regulations, certifications, franchises, permits, licenses, directions and requirements of

- all Governmental Authorities, which now or at any time may be applicable to Borrower, a violation of which could be reasonably expected to have a Material Adverse Effect.
- 5.5 Taxes and Other Liabilities. Borrower will pay and discharge prior to the date on which penalties attach thereto all taxes, assessments and governmental charges, license fees and levies upon or with respect to Borrower, and upon the income, profits and property of Borrower, unless and to the extent that such taxes, assessments, charges, license fees and levies are being contested in good faith and by appropriate proceedings diligently conducted by Borrower, and provided that such reserve or other appropriate provisions as are required in accordance with GAAP will have been made therefor.
- 5.6 Insurance . Borrower will maintain insurance with insurance carriers which Borrower reasonably believes are financially sound on such property, against such risks, and in such amounts as is customarily maintained by similar businesses, naming the Lender as an additional insured as its interests may appear, and file with Lender within five days after Lender's written request therefor a detailed list of such insurance then in effect, stating the names of the carriers, the policy numbers, the insureds thereunder, the amounts of insurance, the dates of expiration thereof and the property and risks covered thereby.
- 5.7 Payment of Indebtedness and Performance of Obligations. Borrower will pay and discharge promptly all lawful Indebtedness, obligations and claims for labor, materials and supplies or otherwise which, if unpaid, could (i) have a Material Adverse Effect or (ii) become a Lien on its property (except as otherwise permitted by this Agreement), provided that Borrower will not be required to pay and discharge or cause to be paid and discharged any such Indebtedness, obligation or claim so long as the validity thereof is being contested in good faith and by appropriate proceedings diligently conducted by Borrower, and that such reserve or other appropriate provisions as are required by the Accountants in accordance with GAAP will have been made therefor.
- 5.8 Use of Proceeds . The proceeds of the Loan shall be used only for working capital for the Business and refinancing of debt.
- 5.9 Maintenance of Property . Borrower shall (i) maintain, keep and preserve all of its properties in good repair, working order and condition, reasonable wear and tear excepted, and from time to time make all necessary and proper repairs, renewals, replacement and improvements thereto, and (ii) maintain, preserve and protect all franchises, licenses, copyrights, patents and trademarks material to its Business, so that the Business carried on in connection therewith may be properly and advantageously conducted at all times.
- 5.10 Conduct of Business . Borrower shall (i) engage in the Business as its principal business, (ii) preserve, renew and keep in full force and effect all of its material contracts, except where it is in Borrower's best interest to terminate or modify any such contract, and (iii) comply in all material respects with the terms of all instruments which evidence, secure or govern indebtedness of Borrower and all Governmental Requirements.

- 5.11 Authorizations . Borrower shall obtain, make and keep in full force and effect all authorizations from and registrations with Governmental Authorities that may be required for the validity and enforceability of this Agreement, the Note, and the documents and instruments executed in connection therewith against Borrower.
- 5.12 Notification of Events of Default and Adverse Developments . Borrower shall promptly (and in any event within there Business Days) notify Lender of the occurrence of (i) any Incipient Default or Event of Default hereunder; (ii) any event, development or circumstance whereby any financial statements most recently furnished to Lender fail in any material respect to present fairly, in accordance with GAAP, the financial condition and operating results of Borrower as of the date of such financial statements; (iii) any material litigation or proceedings that are instituted or threatened (to the knowledge of Borrower) against Borrower, or any of its respective assets; (iv) each and every event which would be an Event of Default (or an event which with the giving of notice or lapse of time or both would be an Event of Default) under any Indebtedness of Borrower exceeding one hundred thousand dollars (\$100,000), such notice to include the names and addresses of the holders of such indebtedness and the amount thereof; and (v) any other development in the business or affairs of Borrower if the effect thereof involves a significant risk of a Material Adverse Effect, in each case describing the nature thereof and the action Borrower proposes to take with respect thereto.
- 5.13 Further Assurances . Borrower shall execute, acknowledge and deliver any and all such further assurances and other deeds, agreements or instruments, and take or cause to be taken all such other action, as shall be requested by Lender from time to time in order to give full effect to this Agreement and the Note and to maintain, preserve, safeguard and continue at all times all or any of the rights, remedies, powers and privileges of Lender under this Agreement and the Note, all without any cost or expense to Lender. Specifically, Borrower will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of common stock to such number of shares as shall be sufficient for the purpose of effecting such conversions of debt, preferred stock or other convertible instruments, or issuing common stock upon the exercise of outstanding warrants or options or other such instruments, including, without limitation, engaging in best efforts to obtain the requisite stockholder approval or any necessary amendments to its Certificate of Incorporation or other action.
- 5.14 Borrower Security Agreement. Borrower will not create, permit or suffer to exist, and will defend the Collateral against and take such other action as is necessary to remove, any Lien on the Collateral (other than the security interests created hereunder or security interests identified on Schedule 5.14 and in existence as of the date hereof), and will defend the right, title and interest of Lender in and to any of Borrower's rights under the Collateral against the claims and demands of all Persons whomsoever.

- 5.15 Further Identification of Collateral . Borrower will if so requested by Lender furnish to Lender, as often as Lender reasonably requests, statements and schedules further identifying and describing the Collateral as Lender may reasonably request, all in reasonable detail.
- 5.16 Reduction of Wainwright Facility. Borrower will pay down the present \$6,000,000 Wainwright Facility to \$5,900,000 or less on or before July 31, 2002 and continue to reduce the Wainwright Facility by no less than \$100,000 each month thereafter such that by December 31, 2002 there is no more than \$5,400,000 outstanding, on December 31, 2003 there is no more than \$4,200,000 outstanding, etc. until all amounts have been repaid to the Bank.

ARTICLE 6 Negative Covenants

Borrower covenants and agrees that, unless the Lender otherwise agrees in writing, as long as any of the Loan remains outstanding, Borrower will not:

- 6.1 Indebtedness. Create, incur, assume or suffer to exist any liability for Indebtedness except for (i) Indebtedness under this Agreement or the Note; (ii) Indebtedness constituting account or trade payables incurred in the ordinary course of business; (iii) Indebtedness outstanding as of the date hereof, provided, however that the terms of any such Indebtedness may not be amended or modified following the date hereof; (iv) Indebtedness not to exceed two hundred fifty thousand dollars (\$250,000) in the aggregate at any time secured by purchase money liens; (v) the Wainwright Facility; (vi) the Reed Facility (vii) the O'Donnell Facility, and (viii) any indebtedness incurred to repay the Loan in full.
- 6.2 Liens . Create, incur, assume or suffer to exist any Lien upon any of its property or assets (including, without limitation, the Collateral), whether now owned or hereafter acquired except for (i) Liens for taxes, assessments or similar charges incurred in the ordinary course of business and not delinquent or being contested in accordance with Section 5.5 or Section 5.7, (ii) mechanics', carriers', workmen's, repairmen's or other like statutory liens incurred by Borrower in the ordinary course of business, provided that the obligations secured thereby are not past due, (iii) Liens granted to Lender pursuant to this Agreement; (iv) liens for purchase money obligations not to exceed two hundred fifty thousand dollars (\$250,000); (v) liens in favor of Wainwright Bank or its successors pursuant to the indebtedness described on Schedule 6.1, and (vi) Liens securing the Reed Facility.
- 6.3 Merger or Acquisition . Consolidate or merge into or with any Person or acquire all or substantially all of the stock, property or assets of any Person; provided, however, that SCH and/or CCN may merge or consolidate into CareCentric, or any wholly-owned subsidiary of CareCentric.

- 6.4 Sale and Exchange of Assets . Sell, exchange, lease, assign, transfer or otherwise dispose of any asset exceeding one hundred thousand dollars (\$100,000) in value to any Person (except in the ordinary course of business or to the extent such asset is obsolete or no longer used or useful in the business of Borrower), or sell, exchange, lease, assign, transfer or otherwise dispose of all or substantially all of its property; provided, however, that SCH and/or CCN may sell, exchange, lease, assign, transfer or otherwise dispose of all or substantially all of its property to CareCentric, or any wholly-owned subsidiary of CareCentric.
- 6.5 Contingent Liabilities. Directly or indirectly create or become or be liable with respect to any Contingent Obligation, except Contingent Obligations arising from the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business.
- 6.6 Investments, Loans, Etc. At any time purchase or otherwise acquire, hold or invest in the stock of, or any other equity interest in, any Person, or make any loan or advance to, or enter into any arrangement for the purpose of providing funds to, or make any other investment, whether by way of capital contribution or otherwise, in, to or with any Person (all of which are referred to as "Investments"), except:
- (a) Investments in short-term certificates of deposit, time deposits and bankers' acceptances issued by any U.S. commercial bank having undivided capital and surplus exceeding \$100,000,000;
- (b) Investments in short-term direct obligations of the United States or obligations of agencies of the United States which are guaranteed by the United States;
- (c) Investments in short-term commercial paper and corporate obligations rated A-1 by Standard & Poor's Corporation or P-I by Moody's Investors Services, Inc.; and
- (d) Loans to Borrower's employees from 401(k) employee benefit plan funds for which Borrower is deemed by law to be the lender.
- 6.7 Fundamental Business Changes . Make any fundamental change in its business activities.
- 6.8 Restricted Payments . Make any distributions to its shareholders, declare or pay any dividends or apply any of its property to the voluntary purchase, redemption or other retirement of, or set apart any sum for the voluntary payment of any dividends on, or make any other distribution by reduction of capital or otherwise in respect of, any shares of its present or future issues of stock (each a "Restricted Payment").
- 6.9 Transactions with Shareholders and Affiliates. Except as set forth on Schedule 6.9, directly or indirectly enter into or permit to exist any material transaction (including, without limitation, the purchase, sale, lease or exchange of any property or the rendering of any service) with any Affiliate of Borrower, except for transactions in the ordinary course of and pursuant to the reasonable requirements of the business of Borrower and upon fair and reasonable

terms which are fully disclosed to Lender and are no less favorable to Borrower or such Subsidiary than would be obtained in a comparable arm's length transaction with a Person that is not an Affiliate of Borrower.

6.10 Changes Relating to Indebtedness . Amend the terms of any Indebtedness (other than Indebtedness under this Agreement or the Note), if the effect of such amendment is to: (i) increase the interest rate on such Indebtedness or any premiums or penalties with respect thereto; (ii) change the dates upon which payments of principal or interest are due on such Indebtedness; (iii) change any default or event of default with respect to such Indebtedness; (iv) change the redemption, defeasance or prepayment provisions of such Indebtedness; (v) change the subordination provisions thereof (or the subordination terms of any guaranty thereof); (vi) change the method of payment of interest thereon; or (vii) change or amend any other term if such change or amendment would materially increase the obligations of the obligor or confer additional material rights on the holder of such Indebtedness in a manner adverse to Borrower or Lender.

ARTICLE 7 Events of Default

- 7.1 Events of Default . Each of the following shall constitute an Event of Default under this Agreement:
- (a) Borrower shall fail to pay when due any payment of principal or interest or any other sum payable hereunder, provided that, for the first two such failures only, Borrower shall be entitled to a five (5) day grace period for the payment of interest;
- (b) Borrower shall fail to comply with any agreement contained in Article 5 or Article 6 (other than Section 6.6 as to which Section 7.1(c) will apply);
- (c) Borrower shall default in the performance of any of its material agreements under any provision of this Agreement or any other agreement between Borrower and Lender;
 - (d) (Intentionally Omitted)
- (e) Borrower or any of its subsidiaries shall institute a voluntary case seeking liquidation or reorganization under Chapter 7 or Chapter 11, respectively, of the United States Bankruptcy Code, or shall consent to the institution of an involuntary case thereunder against it; or Borrower or any of its subsidiaries shall file a petition initiating or shall otherwise institute any similar proceeding under any other applicable federal or state law, or shall consent thereto; or Borrower or any of its subsidiaries shall apply for, or by consent or acquiescence there shall be an appointment of a receiver, liquidator, sequestrator, trustee or other officer with similar powers; or Borrower or any of its subsidiaries shall make an assignment for the benefit of creditors; or

Borrower or any of its subsidiaries shall admit in writing its inability to pay its debts generally as they become due; or, if an involuntary case shall be commenced seeking the liquidation or reorganization of Borrower or any of its subsidiaries under Chapter 7 or Chapter 11, respectively, of the United States Bankruptcy Code, or any similar proceeding shall be commenced against Borrower or any of its subsidiaries under any other applicable federal or state law, and (i) the petition commencing the involuntary case is not timely controverted; or (ii) the petition commencing the involuntary case is not dismissed within 30 days of its filing; or (iii) an interim trustee is appointed to take possession of all or a portion of the property, to operate all or any part of the business of Borrower or any of its subsidiaries, or both; or (iv) an order for relief shall have been issued or entered therein; or (v) a decree or order of a court having jurisdiction in the premises for the appointment of a receiver, liquidator, sequestrator, trustee or other officer shall have been entered therein;

- (f) Borrower shall default in the payment when due of any monetary obligation (other than the Loan) in excess of two hundred fifty thousand dollars (\$250,000);
- (g) One or more judgments against Borrower or attachments against its property, which in the aggregate exceed two hundred fifty thousand dollars (\$250,000), or the operation or result of which could be to interfere materially and adversely with the conduct of the business of Borrower, remain unpaid, unstayed on appeal, undischarged, unbonded and undismissed for a period of 30 days;
- (h) A change in control of Borrower shall occur. For purposes of this subsection 7.1(h), "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of Borrower, whether through ownership of voting securities, by contract, or otherwise; or
- (i) Lender does not have or ceases to have a valid and perfected first priority security interest in the Collateral, subject only to the prior security interests identified on Schedule 5.14.
- 7.2 Termination of Obligations and Acceleration . If any Event of Default described in Section 7.1(a), (e) or (h) shall occur, all Loans shall become immediately due and payable, all without notice of any kind, and Lender shall have no obligation to make further Loans hereunder. If any Event of Default described in Section 7.1 (other than an Event of Default described in Section 7.1(a), (e) or (h)) shall occur, upon ten (10) days notice, and in the case of 7.1(i), upon two (2) days notice, Lender may declare all Loans to be due and payable, whereupon all Loans shall immediately become due and payable, and Lender shall have no obligation to make further Loans hereunder. Any such declaration made pursuant to this Section 7.2 may be rescinded by Lender.
 - 7.3 Other Remedies . If any Event of Default shall have occurred and

be continuing, Lender may exercise in respect of the Collateral, in addition to all other rights and remedies provided for herein or otherwise available to it, all the rights and remedies of a secured party on default under the UCC (whether or not the UCC applies to the affected Collateral) and also may: (a) require Borrower to, and Borrower hereby agrees that it will, at its expense and upon request of Lender forthwith, assemble all or part of the Collateral as directed by Lender and make it available to Lender at a place to be designated by Lender which is reasonably convenient to both parties; (b) with five (5) days written notice, enter upon any premises of Borrower and take possession of the Collateral; and (c) without notice except as specified below, sell the Collateral or any part thereof in one or more parcels at public or private sale, at any of Lender's offices or elsewhere, at such time or times, for cash, on credit or for future delivery, and at such price or prices and upon such other terms as Lender may deem commercially reasonable. Borrower agrees that, to the extent notice of sale shall be required by law, at least ten days' notice to Borrower of the time and place of any public sale or the time after which any private sale is to be made shall constitute reasonable notification. At any sale of the Collateral, if permitted by law, Lender may bid (which bid may be, in whole or in part, in the form of cancellation of indebtedness) for the purchase of the Collateral of any portion thereof for the account of Lender (on behalf of Lender). Lender shall not be obligated to make any sale of Collateral regardless of notice of sale having been given. Lender may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned. To the extent permitted by law, Borrower hereby specifically waives all rights of redemption, stay or appraisal which it has or may have under any law now existing or hereafter enacted.

If an Event of Default has occurred and is continuing, Borrower hereby irrevocably authorizes and empowers Lender to assert, either directly or on behalf of Borrower, any claims Borrower may have, from time to time, against any other party to the Assigned Agreements or to otherwise exercise any right or remedy of Borrower under the Assigned Agreements (including, without limitation, the right to enforce directly against any party to an Assigned Agreement all of Borrower's rights thereunder, to make all demands and give all notices and to make all requests required or permitted to be made by Borrower under the Assigned Agreements).

Beyond the safe custody thereof, Lender shall have no duty with respect to any Collateral in its possession or control (or in the possession or control of any agent or bailee) or with respect to any income thereon or the preservation of rights against prior parties or any other rights pertaining thereto. Lender shall be deemed to have exercised reasonable care in the custody and preservation of the Collateral in its possession if the Collateral is accorded treatment substantially equal to that which it accords its own property. Lender shall not be liable or responsible for any loss or damage to any of the Collateral, or for any diminution in the value thereof, by reason of the act or omission of any warehouseman, carrier, forwarding agency, consignee or other agent or bailee selected by Lender in good faith.

7.4 Application of Proceeds . Upon the occurrence and during the continuance of an Event of Default, the proceeds of any sale of, or other realization upon, all or any part of the Collateral shall be applied: first, to all fees, costs and expenses incurred by Lender with respect to the Loan Agreement; second, to all fees due and owing to Lender; third, to accrued and unpaid interest on the Obligations; fourth, to the principal amounts of the Obligations outstanding; fifth, to any other indebtedness or obligations of Borrower owing to Lender.

ARTICLE 8 Security Interest

- 8.1 Grant of Security Interest . Borrower hereby grants to Lender a continuing security interest in and to all right, title and interest of Borrower in the Collateral, whether now owned or existing or hereafter acquired or arising regardless of where located, to secure payment and performance of the Obligations. Anything herein to the contrary notwithstanding: (a) Borrower shall remain liable under the contracts and agreements included in the Collateral to the extent set forth therein to perform all of its duties and obligations thereunder to the same extent as if this Agreement had not been executed; (b) the exercise by Lender of any of the rights hereunder shall not release Borrower from any of its duties or obligations under the contracts and agreements included in the Collateral; and (c) Lender shall not have any obligation or liability under the contracts and agreements included in the Collateral by reason of this Agreement, nor shall Lender be obligated to perform any of the obligations or duties of Borrower thereunder or to take any action to collect or enforce any claim for payment assigned hereunder. The security interest granted hereby secures the payment and performance of the obligations, liabilities and indebtedness of every nature of Borrower to Lender now or hereafter existing under this Agreement or the Note and all renewals, extensions, restructurings and refinancing thereof, including, without limitation, the principal amount of all debts, claims and indebtedness, accrued and unpaid interest (including, without limitation, interest which but for the filing of a petition in bankruptcy, would accrue on such obligations), the Wainwright Guaranty, and all fees, costs and expenses now or from time to time owing, due or payable.
- 8.2 Affirmative Covenants of Borrower . Borrower shall: (a) do all acts that may be necessary to maintain, preserve and protect the Collateral; (b) pay promptly when due all taxes, assessments, charges, encumbrances and liens now or hereafter imposed upon or affecting the Collateral; (c) procure, execute and deliver from time to time any endorsements, assignments, financing statements and other writings necessary or appropriate to perfect, maintain and protect Lender's security interest hereunder and the priority thereof and to deliver promptly to Lender all records of (1) Collateral or (2) insurance proceeds; (d) appear in and defend any action or proceeding which may affect its title to or Lender's interest in the Collateral; (e) if Lender gives value to enable Borrower to acquire rights in or the use of any Collateral, use such value for such purpose; (f) keep accurate and complete records of the Collateral and provide Lender with such books, records and such other reports and

information relating to the Collateral as Lender may reasonably request from time to time; (g) when an event of default under this Agreement has occurred and after demand, account fully for and immediately deliver to Lender in the form received, all Collateral and all proceeds, endorsed to Lender as appropriate, and unless so delivered all Collateral and all such proceeds shall be held by Borrower in trust for Lender, separate from all other property of Borrower and identified as the property of Lender; (h) keep the Collateral in good condition and repair; (i) at any reasonable time, upon demand by Lender, exhibit to and allow inspection by Lender (or persons designated by Lender) of the Collateral; (j) keep the Collateral (and books and records concerning the Collateral) at the locations(s) set forth in Schedule 8.2 and not remove the Collateral from such location(s) without (1) the prior written consent of Lender or (2) creating a similar security interest at the new location of the Collateral; (k) give thirty (30) days prior written notice of any change in Borrower's chief place of business or trade name(s) or style(s) set forth therein; (1) comply with all laws, regulations and ordinances relating to the possession, operation, maintenance and control of the Collateral; (m) execute and file such financing or continuation statements, or amendments thereto, and such other instruments or notices, as may be necessary or desirable, or as Lender may request, in order to perfect and preserve the security interest granted or purported to be granted hereby under the laws of any applicable jurisdiction, and (n) upon Lender's request, appear in and defend any action or proceeding that may affect Borrower's title to or Lender's security interest in the Collateral.

- 8.3 Negative Covenants of Borrower . Borrower shall not, without the prior written consent of Lender: (a) Use or permit the Collateral to be used unlawfully or in violation of any provision of this Agreement, or any applicable statute, regulation or ordinance or any policy of insurance covering the Collateral; (b) Execute a financing statement covering the Collateral except in favor of Lender, except as permitted under Sections 6.1 and 6.2; (c) Encumber, lease, rent, sell or dispose in bulk, the Collateral or any interest therein; (d) Cause any waste or unusual or unreasonable depreciation of the Collateral; or (e) After default under this Agreement and upon demand, modify, waive or release any provisions of any Account, Contract Right, item of Chattel Paper, Instrument or other right to the payment of money constituting Collateral.
- 8.4 Insurance . Upon execution of this Agreement and all related documents and agreements, Borrower shall insure the Collateral, with Lender named as a loss payee and additional insured as its interests may appear, in reasonable form and amounts, with companies reasonably acceptable to Lender, and against normal risks and liabilities. Borrower shall deliver copies of such policies to Lender at its request. In the event of loss of insured Collateral, Lender may make any claim thereunder, and until the Collateral is promptly replaced by Borrower from the segregated proceeds of the insurance, Lender may collect and receive payment of and endorse any instrument in payment of loss, and apply such amounts received, at Lender's election, to replacement of Collateral or to the Obligations. Lender shall not by the fact of approving, disapproving, accepting, preventing, obtaining or failing to obtain any

insurance, incur any liability for or with respect to the amount of insurance carried, the form or legal sufficiency of insurance contracts, solvency of insurance companies, or payment or defense of lawsuits; Borrower hereby expressly assumes full responsibility therefor and all liability, if any, with respect thereto.

- 8.5 Perfection. Borrower represents and warrants this Agreement creates a valid, perfected and first priority security interest in the Collateral, subject only to the prior security interests identified on Schedule 5.14, securing the payment of the Obligations, and all filings and other actions necessary or desirable to perfect and protect such interest have been duly taken.
- 8.6 Expenses . Lender may incur expenses in connection with the retaking, holding or preparing for sale of the Collateral including, with limitation, reasonable attorneys' fees, appraisal fees, auction fees and advertising costs, and in connection with protecting or enforcing its rights under this Agreement including, but not limited to, reasonable attorneys' fees, which expenses Borrower shall pay and are Obligations secured hereby.
- 8.7 Waivers. Borrower waives any right to require Lender to proceed against any person or to exhaust any Collateral or to pursue any remedy available to Lender. Borrower waives any defenses it may have arising from Lender's failure to perfect or maintain a perfected security interest in the Collateral.
- 8.8 Termination of Security Interests; Release of Collateral . Upon payment in full of all Obligations, the security interest created hereby shall terminate. Upon such termination of the security interest or release of any Collateral, Lender will, at the expense of Borrower, execute and deliver to Borrower such documents as Borrower shall reasonably request to evidence the termination of the security interest or the release of such Collateral which has not yet theretofore been sold or otherwise applied or released. Such release shall be without warranty or recourse to Lender, except as to the absence of any prior assignments by Lender on behalf of its interest in the Collateral, as the case may be.
- 8.9 Cumulative Rights . All rights and remedies of Lender under this Agreement are in addition to all rights and remedies given to Lender contained in any other agreement, instrument or document or available to Lender at law or in equity. All such rights and remedies are cumulative and not exclusive and may be exercised successively or concurrently. No exercise of any right or remedy shall be deemed an election of remedies and preclude exercise of any other right or remedy.

ARTICLE 9 Conversion

9.1 Shares to be Issued . In the event Lender makes an election to convert all or a portion of the unpaid interest and principal on the Loan pursuant to

- Section 2.5 above, CareCentric shall issue to the Lender such number of fully paid and nonassessable shares of Common Stock as is determined by dividing the dollar amount of the Loan designated by the Lender to be converted by the Conversion Price applicable to each such share, determined as hereinafter provided, in effect on the date that the Lender makes his election to convert. The price at which shares of Common Stock shall be deliverable upon such conversion (the "Conversion Price") shall initially be equal to one dollar (\$1.00) per share of Common Stock. Such initial Conversion Price shall be subject to adjustment as hereinafter provided.
- 9.2 Mechanics of Conversion . Before Lender shall be entitled voluntarily to convert the Loan into shares of Common Stock, it shall give written notice to CareCentric at such office that it elects to convert the same and shall state therein the number of shares to be converted and the name or names in which it wishes the certificate or certificates for shares of Common Stock to be issued. CareCentric shall, as soon as practicable thereafter, issue and deliver at such office to Lender, a certificate or certificates for the number of shares of Common Stock to which it shall be entitled. Such conversion shall be deemed to have been made immediately prior to the close of business on the date of Lender's notice of conversion, and the person or persons entitled to receive the shares of Common Stock issuable upon such conversion shall be treated for all purposes as the record holder or holders of such shares of Common Stock on such date.
- 9.3 Adjustments to Conversion Price for Certain Diluting Issuances, Splits and Combinations . The Conversion Price shall be subject to adjustment from time to time as follows:
- (a) Special Definitions. For purposes of this Section 9.3, the following definitions apply:
- (1) "Options" shall mean rights, options, or warrants to subscribe for, purchase or otherwise acquire either Common Stock or Convertible Securities (defined below).
- (2) "Convertible Securities" shall mean any evidences of indebtedness, shares (other than Common Stock) or other securities convertible into or exchangeable for Common Stock.
- (3) "Additional Stock" shall mean all shares of Common Stock issued by CareCentric after the date of this Agreement, and all shares of Common Stock issuable pursuant to Options and Convertible Securities issued by CareCentric after the date of this Agreement, other than (i) shares of Common Stock for which adjustment of the Conversion Price is made pursuant to Section 9.3(d) or 9.3(e) below and (ii) shares of Common Stock that may be issuable upon conversion of CareCentric's Series E Preferred Stock and not held or issuable to John R. Festa pursuant to that certain Stock Grant Agreement dated January 23, 2002 between CareCentric and John R Festa.
- (b) Adjustments. If CareCentric shall issue, after the date of this Agreement, any Additional Stock without consideration or for a consideration per share less than the Conversion Price in effect immediately prior to the issuance of such Additional Stock, the Conversion Price in effect immediately prior to each such issuance shall forthwith be adjusted downward to a price equal to the

price paid per share for such Additional Stock.

- (c) Determination of Consideration. For purposes of this Section 9.3, the consideration received by CareCentric for the issuance of any Additional Stock shall be computed as follows:
 - (1) Cash and Property. Such consideration shall:
- (A) insofar as it consists of cash, be computed at the aggregate amount of cash received by CareCentric excluding amounts paid or payable for accrued interest or accrued dividends;
- (B) insofar as it consists of property other than cash, be computed at the fair value thereof at the time of such issue, as mutually determined in good faith by CareCentric's Board of Directors and the holders of a majority of the Series B Preferred Stock; and
- (C) in the event Additional Shares of Common Stock are issued together with other shares or securities or other assets of CareCentric for consideration which covers both, be the proportion of such consideration so received, computed as provided in clauses (A) and (B) above, as mutually determined in good faith by CareCentric's Board of Directors and the holders of a majority of the Series B Preferred Stock;
- (2) Options and Convertible Securities. The consideration per share received by CareCentric for Additional Stock deemed to have been issued pursuant to this Section 9.3 relating to Options and Convertible Securities shall be determined by dividing:
- (A) the total amount, if any, received or receivable by CareCentric as consideration for the issuance of such Options or Convertible Securities, plus the minimum aggregate amount of additional consideration (as set forth in the instruments relating thereto, without regard to any provision contained therein designed to protect against dilution) payable to CareCentric upon the exercise of such Options or the conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities by
- (B) the maximum number of shares of Common Stock (as set forth in the instruments relating thereto, without regard to any provision contained therein designed to protect against dilution) issuable upon the exercise of such Options or conversion or exchange of such Convertible Securities, or in the case of Options for Convertible Securities, upon the exercise of such Options for Convertible Securities and the conversion or exchange of such Convertible Securities.
- (d) Adjustments to Conversion Prices for Stock Dividends and for Combinations or Subdivisions of Common Stock. In the event that CareCentric at any time or from time to time after the date of this Agreement shall declare or pay any dividend on the Common Stock payable in Common Stock or in any right to acquire Common Stock for no consideration, or shall effect a subdivision of the outstanding shares of Common Stock into a greater number of shares of Common

Stock (by stock split, reclassification or otherwise), or in the event the outstanding shares of Common Stock shall be combined or consolidated, by reclassification or otherwise, into a lesser number of shares of Common Stock, then the Conversion Price in effect immediately prior to such event shall, concurrently with the effectiveness of such event, be proportionately decreased or increased, as appropriate. In the event that CareCentric shall declare or pay any dividend on the Common Stock payable in any right to acquire Common Stock for no consideration, then CareCentric shall be deemed to have made a dividend payable in Common Stock in an amount of shares equal to the maximum number of shares issuable upon exercise of such rights to acquire Common Stock.

- Adjustments for Reclassification and Reorganization. If the Common Stock issuable hereunder shall be changed into the same or a different number of any other class or classes of stock, whether by capital reorganization, reclassification or otherwise (other than a subdivision or combination of shares provided for in Section 9.3(d) above) the Conversion Price then in effect shall, concurrently with the effectiveness of such reorganization or reclassification, be proportionately adjusted so that the Loan under this Agreement shall be convertible into, in lieu of the number of shares of Common Stock which the Lender would otherwise have been entitled to receive, a number of shares of such other class or classes of stock to the number of shares of Common Stock that would have been subject to receipt by the Lender upon conversion of the Loan immediately before that change.
- (f) No Impairment. CareCentric will not, by amendment of its Certificate of Incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms to be observed or performed hereunder by CareCentric, but will at all times in good faith assist in the carrying out of all the provisions of this Section 9.3 and in the taking of all such action as may be necessary or appropriate in order to protect the Conversion Rights of the Lender against impairment.
- (g) Certificates as to Adjustments. Upon the occurrence of each adjustment readjustment of any Conversion Price pursuant to this Section its expense shall promptly compute such readjustment in accordance with the terms hereof and prepare and furnish to Lender a certificate executed by CareCentric's Chief Executive Officer or Chief Financial Officer setting forth such adjustment or readjustment and showing in adjustment or readjustment is based. detail the facts upon which such shall, upon the written request at any time of the Lender, furnish or cause to be furnished to the Lender a like certificate setting forth (i) such adjustments and readjustments, (ii) the Conversion Price in effect immediately before and after such adjustments and readjustments, and (iii) the number of shares of Common Stock and the amount, if any, of other property which at the time would be received upon the conversion of each dollar of the Loan.
 - (h) Notices of Record Date. In the event that CareCentric shall propose at

- any time: (i) to declare any dividend or distribution upon its Common Stock, whether in cash, property, stock or other securities, whether or not a regular cash dividend and whether or not out of earnings or earned surplus; (ii) to offer for subscription pro rata to the holders of any class or series of its stock any additional shares of stock of any class or series or other rights; (iii) to effect any reclassification or recapitalization of its Common Stock outstanding involving a change in the Common Stock; or (iv) to merge or consolidate with or into any other corporation, or sell, lease or convey all or substantially all of its assets, or to liquidate, dissolve or wind up; then, in connection with each such event, CareCentric shall send to the Lender:
- (1) at least twenty (20) days' prior written notice of the date on which a record shall be taken for such dividend, distribution or subscription rights (and specifying the date on which the holders of Common Stock shall be entitled thereto) or for determining rights to vote, if any, in respect of the matters referred to in (iii) and (iv) above; and
- (2) in the case of the matters referred to in (iii) and (iv) above, at least twenty (20) days' prior written notice of the date when the same shall take place (and specifying the date on which the holders of Common Stock shall be entitled to exchange their Common Stock for securities or other property deliverable upon the occurrence of such event).
- (i) Issue Taxes. CareCentric shall pay any and all issue and other taxes that may be payable in respect of any issue or delivery of shares of Common Stock on conversion hereunder; provided, however, that CareCentric shall not be obligated to pay any transfer taxes resulting from any transfer requested by the Lender in connection with any such conversion.
- (j) Reservation of Stock Issuable Upon Conversion. CareCentric shall at all times reserve and keep available out of its authorized but unissued shares of Common Stock, solely for the purpose of effecting the conversion of the Loan hereunder, such number of its shares of Common Stock as shall from time to time be sufficient to effect the conversion of the Loan hereunder, and if at any time the number of authorized but unissued shares of Common Stock shall not be sufficient to effect the conversion of the Loan hereunder, CareCentric will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of Common Stock to such number of shares as shall be sufficient for such purpose, including, without limitation, engaging in best efforts to obtain the requisite stockholder approval of any necessary amendment to this Certificate.
- (k) Fractional Shares. No fractional share shall be issued upon the conversion of the Loan hereunder. All shares of Common Stock (including fractions thereof) issuable upon conversion of the Loan hereunder, shall be aggregated for purposes of determining whether the conversion would result in the issuance of any fractional share. If, after the aforementioned aggregation, the conversion would result in the issuance of a fraction of a share of Common Stock, CareCentric shall, in lieu of issuing any fractional share, either (i) pay the Lender a sum in cash equal to the fair market value of such fraction on the date of conversion (as determined by the closing price of the Common Stock

on the Nasdaq market on the day prior to conversion) or (ii) round such fractional share up to a whole share.

ARTICLE 10 Registration Rights

- 10.1 Registration Rights . CareCentric covenants and agrees as follows:
 - 10.2 Definitions . For purposes of this Section 10:
- (a) The terms "Act" and "Securities Act" mean the Securities Act of 1933, as amended.
- (b) The term "Form S-3" means such form under the Act as in effect on the date hereof or any registration form under the Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by CareCentric with the SEC.
- (c) The term "Form S-4" means such form under the Act as in effect on the date hereof or any registration form under the Act subsequently adopted by the SEC for corporate combinations and exchange offers which permits inclusion or incorporation of substantial information by reference to other documents filed by CareCentric with the SEC.
- (d) The term "Holder" means any person owning or having the right to acquire Registrable Securities or any permitted transferee or assignee thereof.
- (e) The terms "Exchange Act" and "1934 Act" mean the Securities Exchange Act of 1934, as amended.
- (f) The terms "register", "registered", and "registration" refer to a registration effected by preparing and filing a registration statement or similar document in compliance with the Act, and the declaration or ordering of effectiveness of such registration statement or document.
- (g) The term "Registrable Shares" means (i) the Common Stock issuable or issued upon conversion pursuant to Section 2.5 of this Agreement, and (ii) any Common Stock or other securities issued or issuable in respect of shares referenced in (i) above, upon any stock split, stock dividend, recapitalization, or similar event; excluding in all cases, however, any Registrable Securities sold by a Person in a transaction in which such Person's rights under this Section 10 are not assigned.
 - (h) The term "SEC" means the Securities and Exchange Commission.
- (i) The term "Subsidiary" means, with respect to any Person, any corporation, limited liability company, or partnership of which such Person owns, either directly or through its subsidiaries or affiliates, more than fifty percent (50%) of (i) the total combined voting power of all classes of voting securities in the case of a corporation or (ii) the capital or profit interests therein in the case of a partnership.
- 10.3 Request for Registration . Upon request of the Lender, CareCentric will use its best efforts to file within 45 days of a request from Lender a registration statement with the SEC (utilizing Form S-3 or a successor form

thereto and Rule 415 to the extent available) to register Registrable Shares as requested by the Lender. CareCentric shall not be required to file more than three such registration statements (excluding any registration statement which is delayed pursuant to Section 10.5(e) below and through which the Lender is unable to register eighty percent (80%) or more of the amount of Registrable Shares that Lender originally requested to register in such registration statement), and no such filing shall be made prior to the date which is six months after the date of this Agreement.

- 10.4 CareCentric Registration . If CareCentric at any time proposes to register an offering of its securities under the Securities Act, either for its own account or for the account of or at the request of one or more Persons holding securities of CareCentric, CareCentric will:
- (a) give written notice thereof to the Lender (which shall include a list of the jurisdictions in which CareCentric intends to attempt to qualify such securities under the applicable blue sky or other state securities laws) within 10 days of its receipt of a request from one or more Persons holding securities of CareCentric to register securities, or from its decision to effect a registration of securities for its own account, whichever first occurs; and
- (b) use its best efforts to include in such registration and in any underwriting involved therein, all the Registrable Shares specified in a written request by the Lender made within 30 days after receipt of such written notice from CareCentric, except as set forth in Section 10.5(e) below and subject to the currently existing piggyback rights referenced in Section 10.11.
- 10.5 Obligations of CareCentric . If and whenever pursuant to the provisions of this Section 10 CareCentric effects registration of Registrable Shares under the Securities Act of 1933 and state securities laws, CareCentric shall:
- (a) Prepare and file with the SEC a registration statement with respect to such securities and use its best efforts to cause such registration statement to become and remain effective for a period not to exceed two years after the filing (but which period shall be extended by the duration of any delay periods under clause (e) below);
- (b) Use its best efforts to register or qualify the securities covered by such registration statement under the securities or blue sky laws of such jurisdictions as the Lender shall reasonably request, and do any and all other acts and things which may be necessary or advisable (in the reasonable opinion of Lender) to enable Lender to consummate the disposition thereof; provided, however, that in no event shall CareCentric be obligated to qualify to do business in any jurisdiction where it is not now so qualified or to take any action which would subject it to the service of process in suits other than those arising out of the offer or sale of the securities covered by such registration statement in any jurisdictions where it is not now so subject;
- (c) As promptly as practicable prepare and file with the SEC such amendments and supplements to any registration statement and prospectus used pursuant to or in connection with this Agreement as may be necessary to keep

such registration statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement or for such shorter period as may be required herein; and

(d) Furnish to Lender such number of conformed copies of its registration statement and of each such amendment and supplement thereto (in each case including all exhibits, such number of copies of the prospectus comprised in such registration statement (including each preliminary prospectus and any summary prospectus), in conformity with the requirements of the Securities Act), and such other related documents as Lender may reasonably request in order to facilitate the disposition of the Registrable Shares to be registered.

CareCentric may defer the filing ("Filing") of any

- (e) Anything in this Agreement to the contrary notwithstanding:
- registration statement or suspend the use of a prospectus under a currently effective registration statement under this Agreement at its discretion for "Good Cause." "Good Cause" means either if (1) CareCentric is engaged in active negotiations with respect to the acquisition of a "significant subsidiary" as defined in Regulation S-X promulgated by the SEC under the Exchange Act and the Securities Act which would in the opinion of counsel for CareCentric be required to be disclosed in the Filing; or (2) in the opinion of counsel for CareCentric, the Filing would require the inclusion therein of certified financial statements other than those in respect of CareCentric most recently ended full fiscal year and any preceding full fiscal year, and CareCentric may then, at its option, delay the imposition of its registration obligations hereof until the earlier of (A) the conclusion or termination of such negotiations, or the date of availability of such certified financial statements, whichever is applicable, or (B) 60 days from the date of the registration request.
- (ii) In the event CareCentric has deferred a requested Filing, pursuant to the preceding paragraph, such deferral period shall end if CareCentric registers shares for resale by another stockholder of CareCentric. In the event CareCentric undertakes an underwritten public offering to issue CareCentric securities for cash during any period in which a requested Filing has been deferred or if the registration of which CareCentric gives notice under Section 10.4(a) is for an underwritten public offering to issue the CareCentric securities for cash, CareCentric shall include the Registrable Securities in such underwritten offering subject to (A) the right of the managing underwriters to object to including such shares, (B) Section 10.11, and (C) the condition that the Lender shall cooperate in the registration process in all material respects, including execution by the Lender of the underwriting agreement agreed to by CareCentric and the underwriters.
- (iii) If the managing underwriter elects to limit the number or amount of securities to be included in any registration referenced in the preceding paragraph or in Section 10.4(a), all Persons holding securities of CareCentric (including the Lender) who hold registration rights and who have requested

registration (collectively, the "Security Holders") shall, subject to Section 10.11 hereof, participate in the underwritten public offering pro rata based upon the ratio of the total number or amount of securities to be offered in the offering to the total number or amount of securities held by each Security Holder (including the number or amount of securities which each such Security Holder may then be entitled to receive upon the exercise of any option or warrant, or the exchange or conversion of any security, held by such Security Holder). If any such Security Holder would thus be entitled to include more securities than such Security Holder requested to be registered, the excess shall be allocated among the other Security Holders pro rata in a manner similar to that described in the previous sentence.

- (iv) CareCentric may amend any registration statement to withdraw registration of the Lender's Registrable Shares if Lender fails or refuses to cooperate in full and in a timely manner with all reasonable requests relating to such registration and the public offering generally made by CareCentric, the underwriters (if any), their respective counsel and CareCentric's auditors.
- 10.6 Expenses . Without regard to whether the registration statement relating to the proposed sale of the Registrable Shares is made effective or the proposed sale of such shares is carried out, CareCentric shall pay the fees and expenses in connection with any such registration including, without limitation, legal, accounting and printing fees and expenses in connection with such registration statements, the registration filling and examination fees paid under the Securities Act and state securities laws and the filling fees paid to the National Association of Securities Dealers, Inc. Notwithstanding the foregoing, the Lender shall be responsible for the payment of underwriting discounts and commissions, if any, and applicable transfer taxes relating to the Registrable Shares sold by Lender and for the fees and charges of any attorneys or other advisers retained by Lender.
- 10.7 Indemnification . In the event any Registrable Shares are included in a registration statement under this Section 10:
- (a) To the extent permitted by law, with respect to each registration, qualification, or compliance that has been effected pursuant to this Agreement, CareCentric will indemnify and hold harmless Lender, his legal counsel and accountants (each a "Representative"), and any underwriter (as defined in the Act) for Lender and any controlling Person of such underwriter against any losses, claims, damages, or liabilities (joint or several) to which they may become subject under the Act, the 1934 Act or other federal or state law, insofar as such expenses, losses, claims, damages, or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement, including any preliminary prospectus or final prospectus contained therein, offering circular or other document or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required or allegedly required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by CareCentric of the Act, the 1934 Act, any other federal or state securities law or any rule or regulation promulgated under the Act, the 1934 Act

or any other federal or state securities law; and CareCentric will pay Lender, Lender's Representative, underwriter and any controlling Person of such underwriter or controlling Person any legal or other expenses reasonably incurred by such Person in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, or action if such settlement is effected without the consent of CareCentric (which consent shall not be unreasonably withheld), nor shall CareCentric be liable in any such case for any such loss, claim, damage, liability, or action to the extent that it arises out of or is based upon a Violation that occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by Lender.

(b) To the extent permitted by law, Lender will indemnify and hold harmless CareCentric, each of CareCentric's directors, each of CareCentric's officers who has signed the registration statement, each Person, if any, who controls CareCentric within the meaning of the Act, any underwriter, any other Holder selling securities in such registration statement and any controlling Person of any such underwriter or other Holder, against any losses, claims, damages, or liabilities (joint or several) to which any of the foregoing Persons may become subject, under the Act, the 1934 Act or other federal or state law, insofar as such losses, claims, damages, or liabilities (or actions in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by the Lender expressly for use in connection with such registration; and the Lender will pay any legal or other expenses reasonably incurred by any Person intended to be indemnified pursuant to this subsection, in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Lender, which consent shall not be unreasonably withheld; provided, that, in no event shall any indemnity under this subsection exceed the net proceeds after unreimbursed expenses and commissions from the offering received by Lender.

(c) Promptly after receipt by an indemnified party under this Section of notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section, deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense of such action, with counsel mutually satisfactory to the parties; provided, however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one separate counsel,

with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action, if prejudicial to its ability to defend such action, shall relieve such indemnifying party of its liability to the indemnified party under this Section 10.7 only to the extent that the indemnifying party has been injured by the delay. The omission so to deliver written notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section.

(d) If the indemnification provided for in this Section is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage, or expense referred to therein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage, or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other in connection with the statements or omissions that resulted in such loss, liability, claim, damage, or expense as well as any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information, and opportunity to correct or prevent such statement or omission.

- (e) No indemnifying party, in defense of any such claim or litigation, shall, except with the consent of each indemnified party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnifying party of a release from all liability in respect to such claim or litigation.
- (f) To the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with any underwritten public offering are in conflict with the foregoing provisions, the provisions in this Agreement shall control.
- (g) The obligations of the Company and CareCentric under this Section 10.7 shall survive the completion of any offering of Registrable Securities in a registration statement under this Section 10.7, and otherwise.
- 10.8 Information by the Lender . The Lender shall furnish to CareCentric such information regarding the Lender and the distribution proposed by it as CareCentric may reasonably request in writing and as shall reasonably be

required in connection with any registration or qualification referred to in this Section 10.

- 10.9 SEC Rule 144 Reporting and Reports Under Securities Exchange Act of 1934. With a view to making available to the Lender the benefits of SEC Rule 144 promulgated under the Act and any other rule or regulation of the SEC that may at any time permit the Lender to sell securities of CareCentric to the public without registration or pursuant to a registration on Form S-3 or its successor, CareCentric agrees to:
- (a) make and keep public information available, as those terms are understood and defined in Rule 144, at all times from and after ninety (90) days following the effective date of the first registration statement filed by CareCentric for the offering of its securities to the general public;
- (b) take such action, including the voluntary registration of its Common Stock under Section 12 of the 1934 Act, as is necessary to enable the Lender to utilize Form S-3 or its successor for the sale of its Registrable Securities, such action to be taken as soon as practicable after the end of the fiscal year in which the first registration statement filed by CareCentric for the offering of its securities to the general public is declared effective;
- (c) file with the SEC in a timely manner all reports and other documents required of CareCentric under the Act and the 1934 Act after it has become subject to such reporting requirements; and
- (d) furnish to the Lender, so long as the Lender owns any Registrable Securities, forthwith upon request (i) a written statement by CareCentric that it has complied with the reporting requirements of SEC Rule 144 (at any time from and after ninety (90) days following the effective date of the first registration statement filed by CareCentric for an offering of the securities to the general public), the Act and the 1934 Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 or its successor (at any time after it so qualifies), (ii) a copy of the most recent annual or quarterly report of CareCentric and such other reports and documents so filed by CareCentric (at any time after it has become subject to such reporting requirements), and (iii) such other information as may be reasonably requested in availing the Lender of any rule or regulation of the SEC which permits the selling of any such securities without registration or pursuant to such Form S-3 or its successor.
- 10.10 Transfer or Assignment of Registration Rights . The rights to cause CareCentric to register Registrable Securities pursuant to this Section 10 may be transferred or assigned (but only with all related obligations) by the Lender to a transferee or assignee of such securities, provided: (a) CareCentric is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee or assignee and of the securities with respect to which such registration rights are being assigned; (b) such transferee or assignee agrees in writing to be bound by and subject to the terms and conditions of this Agreement; (c) such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Act; and (d) such assignment shall only be effective if it complies with all applicable federal

and state securities laws. For the purposes of determining the number of shares of Registrable Securities held by a transferee or assignee, the holdings of transferees and assignees of a partnership who are partners or retired partners of such partnership (including spouses and ancestors, lineal descendants and siblings of such partners or spouses who acquire Registrable Securities by gift, will or intestate succession) shall be aggregated together and with the partnership.

10.11 Priority and Limitation on Subsequent Registration Rights .

- The parties hereto acknowledge that the rights to registration contained herein shall be subject to (i) the registration rights contained in Section 2(k) of those certain Registration Rights Agreements ("Registration Rights Agreements") dated October 6, 1996 by and among InfoMed Holdings, predecessor in interest to CareCentric) and certain shareholders CareCentric named therein, the registration rights granted pursuant to that certain Second Amended and Restated Agreement and Plan of Merger and Investment Agreement dated as of October 25, 1999 among MCS, Inc., CareCentric, the Lender, Stewart B. Reed and E. Herbert Burk (the "MCS Merger Agreement"), and (iii) the registration rights granted pursuant to that certain Agreement and Plan of Merger dated as of July 12, 1999 among CareCentric Inc., Simione Acquisition Corporation and CareCentric (the "CareCentric Merger Agreement"); provided that the registration rights set forth in the Registration Rights Agreements, the MCS Merger Agreement and the CareCentric Merger Agreement shall only have priority over the registration rights granted pursuant to this Agreement to the extent required in such agreements and to the extent that any such prior rights have not been waived or amended.
- (b) Subject to Section 10.11(d), CareCentric will not grant any right of registration under the Securities Act relating to any of its equity securities to any person or entity other than pursuant to this Agreement unless the Lender shall be entitled to have included in such registration all Registrable Shares requested by Lender to be so included prior to the inclusion of any securities requested to be registered by the persons or entities entitled to any such other registration rights, other than securities subject to the Registration Rights Agreements, the MCS Merger Agreement and the CareCentric Merger Agreement, which shall have priority (but only to the extent that such prior rights have not been waived or amended).
- (c) Subject to Section 10.11(d), for so long as the Lender owns securities representing 20% or more of the voting power of CareCentric on a fully diluted basis, and except as expressly set forth in this Section 10.11, no other Person, other than John E. Reed, shall be entitled to "piggyback" or participate in any of the demand registrations that Lender initiates pursuant to Section 10.3 without such Lender's prior written consent.
- (d) The parties hereto agree that the rights to registration contained herein shall be pari passu with the rights to registration granted in (i) the Reed Facility, and (ii) any outstanding warran held by John E. Reed or his

successors or assigns to purchase common stock of CareCentric.

10.12 Suspension of Registration Rights. The right of any Holder to request registration of shares as provided in this Section 10 shall be suspended during any period of time that all of the Registrable Securities held and entitled to be held (as a result of conversion pursuant to Section 2.5 of this Agreement) by the Lender may immediately be sold under SEC Rule 144.

ARTICLE 11 Miscellaneous

- 11.1 Successors and Assigns . The terms and provisions of this Agreement shall be binding upon, and the benefits thereof shall inure to, the parties hereto and their respective permitted successors and assigns. Neither this Agreement nor any rights or obligations hereunder may be assigned by Borrower without the prior written consent of Lender.
- 11.2 Sale of Interests . Lender is expressly permitted to sell, assign, transfer, negotiate or grant participation in all or any part of or any interest in, its rights and obligations under this Agreement. Except with respect to a transfer to an affiliate of Lender, notice of any such sale, assignment, transfer, negotiation or grant by Lender shall be given to Borrower within a reasonable time period after such event. Upon surrender of the Note at the office of the Borrower, the Borrower shall execute and deliver one or more replacement Notes in the name of the transferee(s) and, if only a part of the Loan is transferred, in the name of Lender.
- 11.3 Lost Promissory Note . Upon receipt of evidence reasonably satisfactory to Borrower of the ownership of and the loss, theft, destruction or mutilation of the Note and indemnification reasonably satisfactory to Borrower or, in the case of any mutilation, upon the surrender of such Note for cancellation to Borrower at its principal office, Borrower at its expense (except as provided below) will execute and deliver to Lender, in lieu thereof a new Note of like tenor, dated so that there will be no loss of interest on such lost, stolen, destroyed or mutilated Note. Borrower may require payment by Lender of a sum sufficient to cover any stamp tax or governmental charge imposed in respect of any such replacement. Any Note in lieu of which any such new Note has been so executed and delivered by Borrower shall not be deemed to be an outstanding Note for any purpose of this Agreement.
- 11.4 No Implied Waiver . No delay or omission to exercise any right, power or remedy accruing to Lender upon any breach or default of Borrower under this Agreement shall impair any such right, power or remedy of Lender, nor shall it be construed to be a waiver of any such breach or default, or an acquiescence therein, or of or in any similar breach or default occurring thereafter, nor shall any waiver of any single breach or default be deemed a wavier of any other

breach or default occurring theretofore or thereafter.

- 11.5 Amendments; Waivers . No amendment, modification, or waiver of or consent with respect to, any provision of this Agreement, shall be effective unless the same shall be in writing and signed and delivered by Lender and Borrower. Any amendment, modification, waiver or consent hereunder shall be effective only in the specific instance and for the specific purpose for which given.
- 11.6 Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall be, only as to such jurisdiction, ineffective to the extent of such prohibition or unenforceability, but all the remaining provisions of this Agreement shall remain valid.
- 11.7 Notices . Any notice which Lender or Borrower may be required or may desire to give to the other party under any provision of this Agreement shall be in writing by overnight delivery service, certified mail, telex or electronic facsimile transmission and shall be deemed to have been given or made when received and addressed as follows:

To Lender:

Mestek, Inc.
R. Bruce Dewey, President
260 North Elm Street
Westfield, Massachusetts 01085
Fax: (413) 568-7428

With a copy to:

J. Nicholas Filler, Vice President, Corporate and Legal Affairs 260 North Elm Street
Westfield, Massachusetts 01085
Fax: (413) 568-7428

If to Borrower, at:

2625 Cumberland Parkway Suite 310 Atlanta, Georgia 30339 Attn: President and CEO Fax: (770) 784-1597

With a copy to:

Sherman A. Cohen Arnall Golden Gregory LLP 2800 One Atlantic Center 1201 West Peachtree Street Atlanta, Georgia 30309-3450 Fax: (404) 873-8631

Any party may change the address to which all notices, requests and other communications are to be sent to it by giving written notice of such address change to the other parties in conformity with this paragraph, but such change shall not be effective until notice of such change has been received by the other parties.

- 11.8 Interpretation . This Agreement, together with the Exhibit to this Agreement, is intended by Lender and Borrower as a final expression of their agreement with respect to the subject matter hereof and is intended as a complete statement of the terms and conditions of such agreement.
- 11.9 No Right of Set Off . Borrower will not be entitled to offset against any of its financial obligations to Lender under this Agreement, any obligation owed to it or any of its Affiliates by or for Lender or any Affiliates of Lender.
- 11.10 Attorneys' Fees and Other Expenses . Borrower further agrees to pay or reimburse Lender for all costs and expenses, including, without limitation, attorneys' fees (including costs of settlement) incurred by Lender after the occurrence of an Event of Default (i) in enforcing the Loan or in foreclosing against the Collateral or exercising or enforcing any other right or remedy available by reason of such Event of Default; (ii) in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of a "work-out" or in any insolvency or bankruptcy proceeding; (iii) in commencing, defending or intervening in any litigation or in filing a petition, complaint, answer, motion or other pleadings in any legal proceeding relating to Borrower and related to or arising out of the transactions contemplated hereby; (iv) in taking any other action in or with respect to any suit or proceeding arising out of this Agreement (bankruptcy or otherwise); (v) in protecting, preserving, collecting, leasing, selling, taking possession of or liquidation of any of the Collateral; or (vi) attempting to enforce or enforcing any security interest in any of the Collateral or any other rights relating to the Obligations.
- 11.11 Governing Law . THE VALIDITY, CONSTRUCTION AND EFFECT OF THIS AGREEMENT AND THE NOTE WILL BE GOVERNED BY THE LAWS OF THE STATE OF DELAWARE, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS. AT THE OPTION OF LENDER, AN ACTION MAY BE BROUGHT TO ENFORCE THE OBLIGATIONS, THIS AGREEMENT, AND/OR THE NOTE IN ANY COURT LOCATED IN THE STATE OF DELAWARE, U.S.A. OR IN ANY OTHER COURT IN WHICH VENUE AND JURISDICTION ARE PROPER.
- 11.12 WAIVER OF JURY TRIAL . TO THE EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND LENDER HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT AND/OR THE NOTE. BORROWER AND LENDER ALSO WAIVE ANY BOND OR SURETY OR SECURITY UPON SUCH BOND WHICH MIGHT, BUT FOR THIS WAIVER, BE REQUIRED OF LENDER. The scope of this waiver is intended to be all-encompassing of any and all disputes that may

be filed in any court and that relate to the subject matter of this transaction, including without limitation, contract claims, tort claims, breach of duty claims, and all other common law and statutory claims. Borrower and Lender each acknowledge that this waiver is a material inducement to enter into a business relationship, that each has already relied on the waiver in entering into this Agreement and that each will continue to rely on the waiver in their related future dealings. Borrower and Lender further warrant and represent that each has reviewed this waiver with its legal counsel, and that each knowingly and voluntarily waives it jury trial rights following consultation with legal counsel. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THE WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS OF THIS AGREEMENT, OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THE LOAN. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

11.13 Indemnification . Borrower will indemnify and hold Lender and its officers, directors, employees, Affiliates, attorneys and agents (collectively, the "Indemnitees") harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims, costs, expenses and disbursements of any kind or nature whatsoever (including without limitation, the reasonable fees and disbursements of counsel) which may be imposed on, incurred by or asserted against such Indemnitees in any manner relating to or arising out of this Agreement or the making of the Loan (collectively, the "Indemnified Matters"); provided, however, that Borrower will have no obligation to an Indemnitee under this Section 11.13 with respect to Indemnified Matters to the extent such Indemnified Matters were caused by or resulted from the gross negligence or willful misconduct of an Indemnitee. Borrower further agrees to pay or reimburse Lender for all costs and expenses, including, without limitation, attorneys' fees (including costs of settlement) incurred by Lender after the occurrence of an Event of Default (i) in enforcing the Loan or in foreclosing against the Collateral or exercising or enforcing any other right or remedy available by reason of such Event of Default; (ii) in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of a "work-out" or in any insolvency or bankruptcy proceeding; (iii) in commencing, defending or intervening in any litigation or in filing a petition, complaint, answer, motion or other pleadings in any legal proceeding relating to Borrower and related to or arising out of the transactions contemplated hereby; (iv) in taking any other action in or with respect to any suit or proceeding arising out of this Agreement (bankruptcy or otherwise); (v) in protecting, preserving, collecting, leasing, selling, taking possession of, or liquidation of any of the Collateral; or (vi) attempting to enforce or enforcing any security interest in any of the Collateral or any other rights relating to the Obligations.

11.14 PROTECTION OF LENDER'S INTEREST AS LENDER. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS AGREEMENT OR ANY OTHER AGREEMENT BETWEEN OR AMONG THE PARTIES AND/OR THEIR AFFILIATES, IN TAKING OR REFRAINING FROM ANY ACTION WITH RESPECT TO THE LOAN OR ANY COLLATERAL, LENDER WILL BE ENTITLED TO

GIVE FULL CONSIDERATION AND WEIGHT TO ITS BEST INTERESTS IN ITS CAPACITY AS A LENDER TO BORROWER WITHOUT TAKING INTO ACCOUNT ANY OTHER OBLIGATIONS, IF ANY, OTHERWISE APPLICABLE TO IT AND BORROWER WAIVES, TO THE FULL EXTENT PERMITTED BY LAW, ANY OBJECTION TO SUCH ACTION OR DECISION BY LENDER, WHETHER BASED ON CONFLICT OF INTEREST OR OTHERWISE. BORROWER EXPRESSLY DISCLAIMS THE EXISTENCE OF ANY FIDUCIARY RELATIONSHIP OR SPECIAL RELATIONSHIP OF TRUST OR CONFIDENCE WITH LENDER. BORROWER FURTHER ACKNOWLEDGES THAT LENDER HAS MADE NO REPRESENTATIONS, WARRANTIES OR COVENANTS TO BORROWER OTHER THAN AS SET FORTH IN THIS AGREEMENT AND LENDER IS NOT OBLIGATED TO AND HAS MADE NO COMMITMENTS WITH RESPECT TO THE PURCHASE OF ANY EQUITY SECURITIES OF THE BORROWER OR THE CONTRIBUTION OF FUNDS TO THE CAPITAL OF BORROWER (OTHER THAN AS SET FORTH IN THE STOCK PURCHASE AGREEMENT OF EVEN DATE HEREWITH) BY VIRTUE OF THIS AGREEMENT.

- 11.15 Counterparts . This Agreement may be executed in any number of counterparts each of which shall be an original with the same effect as if the signatures thereto and hereto were upon the same instrument.
- 11.16 Headings and Sections. Captions, headings and the table of contents in this Agreement are for convenience only, and are not to be deemed part of this Agreement. Unless otherwise specified, references in this Agreement to Sections, Articles, Exhibits or Schedules are references to sections and articles of and exhibits and schedules to, this Agreement.
- 11.17 Joint and Several Liability. All obligations of CareCentric, SCH, and CCN hereunder shall be joint and several. CareCentric, SCH, and CCN acknowledge that the Loan hereunder shall inure to the benefit of all of them. The Obligations shall be deemed to include all joint or individual indebtedness or obligations, contingent or otherwise, of each of CareCentric, SCH, and CCN owed to Lender, which Obligations shall all be secured by the Collateral.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date and year first above written.

as of the date and year first above	written.
LENDER:	Mestek, Inc.
	By_/S/ R.BRUCE DEWEY Its President and COO
BORROWER:	CareCentric, Inc.
	By:_/S/ JOHN FESTA Its President

SC Holding, Inc.

By:_/S/ JOHN FESTA
Its President
CareCentric National, LLC
carecentric national, LLC
By:_/S/ JOHN FESTA
Its President